

# This in an electronic reprint of the original article (publisher's pdf).

Please cite the original article:

Elsanhoury, M., Mäkela, P., Koljonen, J., Välisuo, P., Shamsuzzoha, A., Mantere, T., Elmusrati, M., & Kuusniemi, H. (2022). Precision positioning for smart logistics using ultra-wideband technology-based indoor navigation: A review. *IEEE Access, 10*, 444134–44445. https://doi.org/10.1109/ACCESS.2022.3169267







Received March 21, 2022, accepted April 9, 2022, date of publication April 21, 2022, date of current version April 29, 2022. Digital Object Identifier 10.1109/ACCESS.2022.3169267

## **Precision Positioning for Smart Logistics Using Ultra-Wideband Technology-Based Indoor Navigation: A Review**

MAHMOUD ELSANHOURY<sup>©</sup><sup>1</sup>, (Graduate Student Member, IEEE), PETTERI MÄKELÄ<sup>1,2</sup>, JANNE KOLJONEN<sup>©</sup><sup>1</sup>, PETRI VÄLISUO<sup>©</sup><sup>1</sup>, AHM SHAMSUZZOHA<sup>1</sup>, TIMO MANTERE<sup>1</sup>, MOHAMMED ELMUSRATI<sup>1</sup>, (Senior Member, IEEE), AND HEIDI KUUSNIEMI<sup>©</sup><sup>1,3</sup>, (Member, IEEE)

<sup>1</sup>School of Technology and Innovations, University of Vaasa, FI-65200 Vaasa, Finland
<sup>2</sup>Seinäjoki University of Applied Sciences, 60100 Seinäjoki, Finland
<sup>3</sup>National Land Survey, Finnish Geospatial Research Institute, FI-02430 Masala, Finland

Corresponding author: Mahmoud Elsanhoury (mahmoudessam08@gmail.com)

This work was supported in part by the TULEVA Project, in part by the European Regional Fund, European Union, and in part by the

University of Vaasa, Finland.

**ABSTRACT** Logistics is an important driver for the competitiveness of industries and material supply. The development of smart logistics, powered by precise positioning and communication technologies can significantly improve the efficiency of logistics. The emerging technology of ultra-wideband (UWB) precision positioning has attracted significant attention throughout the previous decade owing to its promising capabilities over other radio frequency-based indoor localisation systems. In addition, UWB is characterised by large bandwidth and data rate, short message length, low transmission power and high penetration capability, which are all favourable for indoor positioning applications. However, UWB localisation technology faces several challenges that are somewhat similar to other technologies, such as mitigating errors that originate from non-line-of-sight (NLOS) situations and tackling signal interference in dense environments, and when required to operate in extreme conditions. This paper reviews the most recent advances made in UWB positioning systems over the last five years, with a focus on high-ranking articles. In addition to going through more conventional solutions to UWB challenges, modern solutions, which involve the use of machine learning and sensor data fusion, are discussed. We highlight the most promising findings of the recently implemented and foreseen UWB positioning systems by providing a summary of each reviewed article. Additionally, we address a major challenge that faces the UWB positioning technology: NLOS situations, focusing on some proposed remedies such as multi-sensor fusion and machine learning. As an application, this study introduces how UWB technology promotes smart logistics by offering indoor positioning to improve efficiencies in the delivery of goods from the source to the customer. Furthermore, it demonstrates the benefits of UWB technology for accurate positioning and tracking of both stationary and moving items, and machinery in an indoor logistics environment.

**INDEX TERMS** Ultra-wideband (UWB), indoor positioning systems (IPS), smart logistics, navigation and localisation, machine learning, sensor fusion.

#### **I. INTRODUCTION**

In recent years, the importance of indoor positioning systems (IPS) has been significantly increasing due to the rapid development and popularization of smart devices and technologies. The extensively used technology for positioning is the global navigation satellite systems (GNSSs) that provide reliable outdoor positioning information. However, due to the

The associate editor coordinating the review of this manuscript and approving it for publication was Peng-Yong Kong<sup>(b)</sup>.

rather weak signals, their performance in the indoor environment is not sufficient. In addition, indoor surroundings are usually complex and varying in nature due to transferable obstacles that cause variation of signal and noise levels [1]. In such situations, substantial research and development of indoor positioning systems are required with alternative wireless technologies such as Wi-Fi [2], geomagnetic field [3], Bluetooth low energy (BLE) [4], dead-reckoning technique [5], ultra-wideband (UWB), and radio signal tags [6]. Precise location information of people and assets is a crucial element for Internet of Things (IoT) and smart logistics applications [7]. The various IPS technologies have been widely adopted by researchers and industrial companies during the past decade due to their broad spectrum of applications, including smart logistics [8], tracking medical equipment in healthcare centres [9], tracking individuals in crowded venues [10], ship-building and offshore industry [11], and construction industry [12].

There are numerous indoor localisation technologies with diverse properties. They are often distinguished by positioning accuracy, system robustness, computational power and cost. When compared to indoor positioning technologies using narrowband signals, such as Bluetooth and Wi-Fi, UWB has many advantages [10], [13].

An UWB signal contains different frequency components due to its wide bandwidth, which increases its probability of penetrating obstacles. In addition, because of its low power spectral density, UWB does not interfere with most other radio systems. Because of its short pulse duration, UWB yields high ranging accuracy and good performance under multipath conditions.

Several technologies have been proposed for obtaining a tracking solution in indoor logistics. The selection of an appropriate technology mainly depends on the required precision, number of assets that need to be tracked and speed of the moving assets. Compared to other available technologies, UWB can provide low-cost accurate positioning within few centimetres accuracy; hence, it is frequently used in industrial environments such as smart factory [14], smart logistics [15], [16], warehouse management [17], vehicle localisation [18], robot positioning [19], and smart city [20]. UWB enables simultaneous real-time tracking of objects and provides their localisation despite indoor obstructions due to the advantage of working under non-line-of-sight (NLOS) conditions [21], [22]. UWB technology enables an efficient combination of accuracy, scalability and reliability, which are very crucial in the applications of indoor logistics, such as tracking people and assets, and controlling the automated guided vehicles (AGVs) [23], [24].

Numerous survey articles were presented throughout the past decade with a focus on addressing UWB as an IPS. For example, in 2016, Alarifi *et.al.* [10] published their comprehensive survey paper that presented UWB-IPS technology with new taxonomies as well as analysing the strengths, weaknesses, opportunities, and threats (SWOT) to form a widely acknowledged state-of-the-art review. Similar approach to Alarifi, was presented by Mazhar *et.al.* [25] in 2017. Another survey article that was focused on IPSs in general and UWB in particular was Yassin *et.al.* [26], in which the authors presented a detailed review about the structure and challenges of UWB-IPS compared to other IPS technologies. More surveys are tacitly cited throughout the article and discussed in the relevant sections.

This paper highlights the advances achieved in the realm of UWB localisation during the past half-decade, with a focus

on the most recently published literature found from IEEE Xplore, Google Scholar and Crossref platforms. The search terms were limited to "Ultra Wideband" AND "Positioning" OR "Localisation" within the last five years. In addition to the newness of the articles, their selection criteria were based on impact, relevance, higher ranking in the Finnish national system [27], novelty and citation score. However, some older articles were also included due to their importance and the unchanged scientific principles that they describe.

Note that the addressing of UWB throughout this article is intended to review UWB as an IPS rather than a communication system. Hence, we focus on presenting the recent momentum of UWB technology from an IPS design perspective.

Throughout the article, a particular focus was given on applying the UWB technology to deploy indoor positioning in smart logistics systems. For smart logistics, indoor positioning plays an essential role by tracking resources, materials and employees in real-time. This location-based indoor navigation enables logistics providers with asset tracking with reduced search times, process automation and optimisation, increased efficiency and safety for employees [17], [28]. In the case of assets tracking, indoor positioning offers realtime seamless tracking of goods, pallets, vehicles and monitoring of goods' conditions, such as temperature, humidity and dew points. For process automation and optimisation, IPS supports automation through precise geo-based task assignments. By providing precise positioning, IPS provides added safety and security in logistics sites with access authorisations and ensures faster evacuation of employees in case of emergencies [23], [24].

The main contribution of this article is to present the most recent advances in the fast-developing UWB precise positioning technology with a focus on smart logistics applications that can embrace the technology to achieve better system enhancements and optimisations. The availability of new UWB chips at affordable prices has catalyzed the development of new algorithms for new application areas, including machine learning, sensor fusion and collaborative positioning. This article dissects the topic thoroughly down to the basic concepts of localisation and algorithms besides keeping a compact style of summarizing the literature to help researchers and industrial firms cope with the recent advances in UWB technology and its foreseen future.

In addition, a major objective of this review is to provide the researchers, practitioners of smart logistics and related fields with a starting point to understand the potential and limitations of UWB indoor navigation. To the best of our knowledge, other existing reviews have not adopted a similar approach. This article explains the fundamentals of positioning techniques and algorithms not only at the detailed mathematical and algorithmic level but also at the conceptual level. Commercially available equipment is presented, as well. Moreover, numerous articles are summarized in several application-specific tables to easily compare the methods and algorithms in the selected application areas. Consequently, also readers who do not have previous knowledge of positioning techniques can utilize this article to obtain the basic knowledge of UWB positioning and its use in smart logistics.

The rest of the article is organized as follows: Section II contains a brief review of existing IPSs and presents the role of IPSs in smart logistics. Section III discusses the advantages and challenges of UWB, licensing and regulations, signal attributes, system architecture, and commercial applications.

Section IV explains the observables and positioning technologies used with UWB. Section V discusses positioning algorithms, such as least-squares methods, closed-form solutions and various kinds of Bayesian filters. In addition, NLOS identification and mitigation methods are explained. Recent advances in UWB positioning literature is stated in Section VI, whereas advances in sensor fusion techniques are elaborated in Section VII. As the last few years have witnessed a surge on the literature on the adoption of artificial intelligence (AI) solutions, various machine learning (ML) approaches for UWB are discussed in Section VIII. A final discussion on indoor positioning is stated in Section IX, while, the paper is concluded in Section X.

#### **II. INDOOR POSITIONING SYSTEMS**

An IPS is a real-time system that uniformly calculates the location of an agent or object, moving or stationary, inside buildings [10], [29]–[31]. An IPS has facilitated numerous navigation applications that require continuous knowledge of the indoor locations of people and objects. It provides assisted navigation for limited-sight and vision-impaired people, tracking of visitors in heavily crowded venues, such as airports and malls, and automated guidance in tourist attractions. Industrial applications use IPSs to ensure precise navigation of industrial robots and accurate positioning of personnel, tools and equipment. Medical applications include monitoring of patients in hospitals and healthcare centres, localisation of crucial medical equipment and enabling navigation for medical robotic assistants [10].

Outdoor positioning has been made feasible owing to GNSSs, such as Global Positioning System (GPS), Galileo, GLONASS and stand-alone cellular positioning systems [31], [32]. However, these systems perform poorly in indoor venues due to the satellite signal deterioration caused by signal attenuation, shadowing and multipath fading imposed by indoor concrete and metallic structures [26], [32]–[34]. Thus, accurate indoor positioning techniques have been proposed to address the deep fading of conventional positioning methods. In addition, hybrid approaches have been proposed to combine multiple indoor and outdoor positioning techniques to mitigate errors and provide more accurate and robust navigation. Numerous indoor positioning techniques can provide positioning accuracy of only a few centimetres as stated in [10], [26], [30]. However, combining multiple indoor positioning techniques requires scrutinised studies to allocate suitable resources for each method and identify the optimal combination as the resource requirements vary from method to method. The optimisation objectives and constraints may concern the energy budget, RF bandwidth budget, infrastructure costs, number of beneficiaries (users) and active time budget.

#### A. SMART LOGISTICS AND MANUFACTURING

Nowadays, smart logistics is considered a fundamental pillar of Industry 4.0. It enables companies to orchestrate several critical activities, such as demand forecasting, sales planning and inventory management [35], [36]. Smart logistics or 'Logistics 4.0' promotes the acceleration of all logistics processes through planning and controlling with smart tools, technologies and methods. The use of smart technologies helps to gather the necessary information to monitor and control the material flow and for many other purposes. Through intelligent positioning technologies, tools and equipment, global supply and logistics chains are becoming increasingly efficient and effective [37]. Such positioning technologies significantly contribute to end-to-end visibility, improvement in product routing as well as control and replenishment of inventories and mobile assets [38]. A generic supply and logistics chain is displayed in Figure 1, which is divided into inbound and outbound supply and logistics chains.

Smart logistics is considered a fairly complex phenomenon that can be easily applied in geographically dispersed areas for tracking raw materials of the highest quality but lowest cost. Such a phenomenon is generally characterised by the use of new technologies, such as IoT, 5G, sensors, radio frequency identification (RFID) tags, smart products, actuators and intelligent machines [38], [39]. It has the ability, reliability, traceability and authenticity of the information, which are useful to establish intelligent contractual relationships among the stakeholders of the supply chains. Moreover, smart logistics includes considerable potential for improving the logistics process through the application of communication and information technologies at all levels of the value chain. Figure 2 displays the fundamental elements and functionalities of the smart supply and logistics chain, showing that it starts from the supplier and eventually moves forward till the end customer through smart transportation, manufacturing, smart warehousing and smart delivery stages.

Figure 2 also shows that, at each stage in the smart supply and logistics chain, several activities are orchestrated. For instance, at the transportation stage, activities such as traceability of items, location tracking and real-time routing are orchestrated and monitored for smoother operations. Tracking logistics items is an essential issue in today's supply chain and inventory management. In addition, finding items both in indoor and outdoor environments is most critical in any supply chain and logistics management [40]. Due to the trend towards autonomous systems, most logistic systems today are operated without the direct involvement of the workforce to control them. In such a changing environment, smart logistics can be helpful to deliver items through various available precise positioning technologies [41], [42].



FIGURE 1. Various stakeholders and the flow of material, assets and information in a generic supply and logistics chain.



FIGURE 2. Fundamental elements and functionalities of smart supply and logistics chain.

Although several technologies are available to provide outdoor positioning towards logistics items, most of them are not suitable for indoor positioning. In indoor environments, such as warehouses and factory floors, precision positioning of components, parts and products can be achieved through available technologies, such as UWB, Wi-Fi, 5G, 3D imaging, sensors and imaging radio signals [25], [43], [44]. The application of such technologies helps to track indoor logistics items, which can help in minimising the time required to locate items and avoid delays due to the wrong location. A precision positioning technique enables the automatic delivery of goods by using uncrewed intelligent vehicles and aerial vehicles (UAVs) to designated locations while reducing environmental influences.

#### B. INDOOR POSITIONING SYSTEMS IN SMART LOGISTICS AND MANUFACTURING

Due to the recent advancements in indoor positioning technologies, a growing interest has emerged to utilize location data in logistics and manufacturing. Location data of assets and materials can be used to improve the efficiency, safety and security of manufacturing operations. Real-time tracking of machines and materials yields new possibilities to improve the production processes and follow the material flows. The authors of [45] divided logistics units into six identification layers: (0) raw material (items), (1) package, (2) transport unit, (3) unit load (pallet), (4) container and (5) transportation unit (e.g. truck, ship and train). GNSSs are typically used for tracking containers and transport equipment (two highest layers). However, the smaller cargo units (layers 1-3) are typically handled indoors, which can be tracked using indoor positioning technologies, such as UWB. However, UWB is still a relatively expensive and power-consuming technology, and RFID is a better technology to track the materials and lowest level packages and items.

Real-time location tracking is increasingly attracting global logistics companies due to the need for visibility. Especially, the application of IPSs in logistics and manufacturing has increased recently [46]. In the case of an indoor environment, such as a warehouse, an IPS contributes to tasks such as minimising the time spent to look for the right pallet, optimising routes and preventing accidents. In the case of smart logistics, companies generally use RFID technology, which can track the inventory and identify goods [42], [47], [48]. However, the limited power source of RFID tags minimises their operational range to a couple of metres. Therefore, they are mainly used for identification rather than positioning purposes. Similarly, Bluetooth and Wi-Fi are also used for indoor positioning, but their operating ranges are no more than 3-5 m [49]. To obtain accurate positioning of logistics items, companies are nowadays exploring UWB-based positioning systems, which enable real-time positioning of goods, assets and people with an accuracy level of 5 to 30 cm. It can provide out-of-the-box localisation with higher ranging accuracy than Wi-Fi or Bluetooth or other active radio solutions [50], [51].

In the case of autonomous robots operating in indoor environments (e.g. warehouses), accurate positioning is essential for navigation. While GNSS-based localisation is unreliable in indoor environments, localisation by UWB technology can accelerate the adoption and ubiquity of distributed robotics systems. Nowadays, UWB-based technology is commonly used in indoor robotic applications from home cleaning to warehouse transportation, including the rapidly emerging autonomous last-mile delivery solutions [46]. In the case of intelligent manufacturing, it is essential to track the parts along the production chain to take the right decisions. Real-time tracking of both stationary and moving parts in the production floor ensures safer operation with a reduced lead time [52], [53]. In addition to parts, it is also essential to track workers' movement on the production floor to enhance operational flexibility. For smart manufacturing, UWB-based location technology is suitable because of its inherent accuracy and reliability [15], [54]. It is considered as the most optimal and accurate approach to ensure indoor localisation [55]. By providing indoor localisation and tracking solutions for vehicles, people and goods, UWB technology promotes increased transparency, safety and productivity in internal logistics on the factory floors [55].

#### C. PERFORMANCE METRICS OF INDOOR POSITIONING SYSTEMS

The performance of localisation technologies can be assessed using a pyramid-like scheme with system accuracy as the baseline, integrity as the second metric, continuity as the third, and availability as the peak paramount [56]. System accuracy is the degree of conformance of the estimated positioning values to the ground truth. The integrity of localisation systems, as defined by [56], is the trustworthiness of the information provided by the navigation engine. Continuity is the probability of the system to maintain the desired service level within the operation period, while availability is the percentage of time in which the navigation engine is up-running for positioning and can be used by its intended users. Accuracy of the estimated position is one of the most important performance metrics for indoor positioning systems. Accuracy is often reported as the error distance between the estimated and actual locations, while a location precision is reported in percentages of position information, which is within the distance of accuracy. The most commonly used metrics of accuracy and location precision are the root mean square error (RMSE), the mean absolute error (MAE), the distance root mean square error and circular error probability.

The accuracy of the location estimate depends on the accuracy of individual measurements and the mutual geometry of the tag and anchors. In the time of arrival (TOA) and time difference of arrival (TDOA) methods, the accuracy of the position is expressed as the product of a geometric factor and a range measurement error factor.

In addition to the above mentioned metrics, there are other performance metrics presented in the literature, such as scalability, cost and privacy [10]. The scalability of IPS describes how many tags the system can support per time unit per geographic area. The cost measures the physical limitations and requirements associated with the implementation of a particular technology in terms of technical and financial resources. Money, power consumption and hardware dimensions are examples of cost metrics. Privacy is a concern in networkcentric systems, where the location estimation takes place in the server. In the self-positioning model the device estimates its own position, and no one else may know where the device is. Coverage was mentioned as an important parameter [10]; however it can also be considered a property of IPS rather than a performance metric.

#### D. INDOOR POSITIONING TECHNOLOGIES

IPSs can use various signal technologies, such as radio frequency, infrared, ultrasonic, inertial, optical and electromagnetic [10], [57]. In addition, the positioning system commonly estimates the location of the target device by fusing measurements of two or more signal technologies. The indoor positioning applications have various requirements in terms of the performance metrics. Thus, the technology should be carefully chosen to satisfy these requirements [10]. For example, the navigation systems of AGVs might require highly accurate and reliable position estimates, but the power consumption or price of the sensor mounted in an AGV is not critical. In contrast, low price and power consumption are required from the tags used to locate people and assets in a warehouse, but the accuracy of the position estimate is less critical than that in AGV applications.

Laser triangulation is commonly used in the indoor navigation of AGVs [58]. The laser positioning system uses a laser scanner mounted on top of the vehicle. The laser scans the mirrors mounted at the known locations in the area and measures the angles between the vehicle and device. The vehicle's position is estimated using triangulation with centimetrelevel accuracy. Another commonly used approach for AGV navigation is to use light detection and ranging (LiDAR) and inertial motion unit (IMU) measurements [59] or

 TABLE 1. Summary of RF-based signal technologies for local positioning systems.

		Typical	
Technology	Accuracy	Range	Method
UWB	2-50 cm	15-50 m	TOA/TDOA trilatera- tion
Wi-Fi RSSI	< 10 m	35 m	Location fingerprint- ing
Wi-Fi CSI	< 5 m	35 m	Location fingerprint- ing, AoA
Wi-Fi RTT	> 1 m	35 m	TW-TOA
Bluetooth RSSI	< 5 m	< 50 m	Location fingerprint- ing
Bluetooth DF	< 1m	< 50 m	AOA, AOD
5G sub-6 GHz	3-10 m	0.5-2 km*	TDOA, TW-TOA, AOA, AOD
5G mmWave	0.2-0.5 m	200 m	TDOA, TW-TOA, AOA, AOD

\* Depends on the frequency and the transmit power

image-based (visual) localisation [60] together with simultaneous localisation and mapping (SLAM) algorithms.

However, the laser and vision-based technologies are relatively expensive, and their energy consumption is too high for the tags required in the applications to locate people, materials and assets. In such applications, radio frequencybased signal technologies are commonly used for positioning. An RF signal is used for positioning for the same reasons as it is used for communication. The most important advantages of RF signals are that they can penetrate obstacles and have a wide communication bandwidth. RF-based IPSs use RFID, UWB, wireless local area network (WLAN), Bluetooth or cellular network signals for location estimation [10]. The position can be estimated from the signals of these systems by using proximity information, trilateration, triangulation or location fingerprinting methods. The properties of these systems are summarised in Table 1.

In the case of RFID, the proximity method is generally used. Location fingerprinting, which provides a roomlevel or couple of metres accuracy, is commonly used with WLAN [61] and Bluetooth [62] received signal strength indicator (RSSI) measurements. RSSI describes only the average attenuation of the signal in the communication channel. More accurate and stable results may be achieved using channel state information (CSI), such as channel impulse response (CIR), which contains more information than a single RSSI value [63]. Some researchers used an antenna array in a Wi-Fi network to estimate CSI, thus, enhancing the accuracy and stability of the fingerprinting positioning system to a few meters [64], [65]. A WLAN signal can also be used for trilateration using the Wi-Fi round trip time (RTT) measurements, referred to as 802.11mc [66]. In Bluetooth direction finding (DF), the target's position is estimated using the triangulation method with the angle of arrival or departure measurements of Bluetooth signals. According to the authors

TABLE 2.	Different	communication	band us	sage scer	narios ar	ıd their
erminolo	gy.			-		

	Fractional	Band Ratio
Band Type	Bandwidth $B_F$	$b_r = \frac{f_h}{f_l}$
Narrowband (NB)	$0.00 < B_F \le 0.01$	$0.00 < b_r \le 1.01$
Wideband (WB)	$0.01 < B_F \le 0.25$	$1.01 < b_r \le 1.29$
Ultra-Wideband (UWB)	$0.25 < B_F < 2.00$	$b_r \ge 1.29$

of [67], Bluetooth DF can achieve accurate measurements if not severely affected by multipath interference. Future 5G NR mmWave technology is expected to provide centimetre-level positioning accuracy and one degree orientation accuracy for the device when the TOA, TDOA and angle of arrival (AOA) are used [68].

UWB is well suited for many positioning applications. In the case of UWB, the position can be estimated with centimetre-level accuracy using triangulation or trilateration positioning methods or both, as discussed in subsequent chapters.

#### **III. UWB POSITIONING**

UWB is a wireless short-range radio technology whose communication channel propagates information over a wide spectrum by modulating either a carrier-based waveform or a carrier-less baseband signal in the form of short-width pulses [69]. According to the Federal Communication Commission (FCC) and the International Telecommunication Union (ITU)-R, UWB possesses a spectrum that occupies a bandwidth greater than 20% of the central frequency or has a bandwidth of at least 500 MHz. A UWB RF signal occupies the ultra 500 MHz bandwidth, which facilitates the transmission of large data sizes upon the consumption of lesser energy than other technologies [10], [31], [70]. To differentiate among narrowband (NB), wideband (WB) and UWB, the FCC classification scheme adopts fractional bandwidth calculation,  $B_F$  which is a dimensionless frequency-independent indicator, calculated using Equation (1) as follows: [70], [71]

$$B_F = 2\left(\frac{f_h - f_l}{f_h + f_l}\right) \tag{1}$$

where  $f_h$ ,  $f_l$  are the higher and lower frequency bands of the signal, respectively. Hence, the band type is determined using the data shown in Table 2 [70].

In 2002, the FCC described UWB technology as an emerging promising technology that holds great advances for various applications [72], such as imaging systems, ground-penetrating radars (GPRs), wall-imaging systems, medical systems, surveillance systems, vehicular radar systems, communications and measurements systems [73]. UWB can transmit high data rates using tiny pulses of the spectrum spread over wider frequency bands with low PSD, which provides the signal higher penetration capability than most RF waves. Moreover, some types of UWB signals (e.g. impulse radio UWB) do not require sinusoidal carrier waves, which

in turn reduces the power required for transmission. The combined advantages of a UWB signal makes it a prominent candidate for real-time applications, such as 1) tracking and navigation, 2) sensor network communications, 3) ranging and imaging, and 4) extremely high-data-rate short-range communication (e.g. wireless UWB).

Recently, UWB has been widely adopted in personal area networks (PANs), precise indoor positioning, indoor tracking and navigation systems. UWB positioning relies on the unique radio frequency characteristics associated with UWB technology to provide accurate estimates for indoor locations based on the TOA, AOA and TDOA of the signal. The UWB positioning signal takes the form of a low-power short-pulse transmission with large bandwidth [10], [31], [70], making it robust, precise and secure.

## A. ADVANTAGES AND CHALLENGES OF UWB COMPARED TO OTHER INDOOR POSITIONING TECHNOLOGIES 1) ADVANTAGES

UWB was first commissioned by the FCC for public use in 2002. Earlier, it was utilized solely by the US military for classified applications [13]. The structure of a UWB signal comprises the transmission of short pulses within large bandwidth ranges between 3.1 and 10.6 GHz, which yields UWB superiority over NB signals. Owing to the large bandwidth and short duty cycle, UWB possesses a larger capacity and higher data rate, which make it a suitable candidate for RF-based IPS implementation. Moreover, UWB lies in the unlicensed spectrum, which can be used by anyone without prior notification. Additionally, the pulse nature of the UWB signal increases its penetration capability. Therefore, UWB tags on mobile targets do not require a direct line of sight (LOS) with its anchors.

However, some scenarios found in dense environments might have negative effects on the UWB signal, causing multipath deterioration and interference with neighbouring frequencies in the spectrum [10]. In addition, the low transmission power can be ineffective in large-sized indoor spaces, as it disallows the signal from travelling to longer distances due to path loss attenuation. Hence, additional UWB anchors are required, which increases costs and complexity [13].

UWB offers numerous benefits over narrowband signals, which widens the range of affected applications. First, UWB is an unlicensed free spectrum that can be used without prior licensing. The UWB spectrum was made free for commercial use in 2002, but before that, it was restricted to military operators, mainly the Department of Defense, for classified applications [13]. UWB has a larger bandwidth than other positioning techniques, ranging from 3.1 to 10.6 GHz [73], [74], which provides it with the superiority in many aspects. For example, based on Shannon's law, the large UWB bandwidth provides large capacity for an RF signal, which implies a high data rate transmission that can support real-time applications, such as instant video streaming [13], [73]. Owing to their large bandwidth, UWB communication



FIGURE 3. Comparison between the attributes of UWB and other various positioning technologies (the dimensions of the shapes are only indicative).

systems are highly robust, operating at higher data rates (110 Mbps) than other RF technologies, making it the highest data rate achieved so far in the precise positioning realm. Another benefit of the large bandwidth is the UWB system's capability of performing in low signal-to-noise-ratio (SNR) communication channels [13], which provides immunity against multipath degradation. The high level of multipath resolution is mainly attributed to the nature of pulse-based RF communication, which occupies the entire bandwidth for each pulse, unlike other carrier-based communications [75], UWB systems do not require a clear LOS, but the UWB communication is perfectly possible under NLOS conditions. However, in positioning applications, NLOS situations might produce erroneous sensor readings, which can disturb the position estimation. Additionally, the short-pulse low-power nature of UWB signals is a major advantage of UWB, making it a suitable candidate for indoor positioning applications, as demonstrated in Figure 3 [26], [76].

Additionally, the UWB signal transmits at low average power due to the short-pulse nature of transmission, submerging it within the noise floor(-40 dBm/MHz), which helps in saving transmitter energy, enhancing the battery life and bestowing resistance against jamming and interception.

#### 2) CHALLENGES

Although UWB technology offers numerous benefits for indoor positioning applications, the technology faces several challenges and drawbacks that affect its performance.

UWB technology is known for its coexistence with other RF systems, but this is not always true. The technical report published by the US National Institute of Standards and Technologies [77] stated that UWB can cause interference to existing nearby RF systems and vice versa. Examples

of the potentially affected RF technologies are the GPS, 3G and WiMAX communication systems [10], [77] due to the misconfiguration of wireless transceiver devices. Many countries have imposed regulations to mitigate the possible interference, which are covered in the following sections.

The low power transmission of UWB is considered an advantage, yet it limits the overall power consumption for the transmitter and receiver. For example, the low-power UWB signal can either travel short distances at a high data rate or long distances at a low data rate [78]; hence, the range of the UWB anchor will be limited. This can only be compensated by using more UWB anchors, which limits the scalability [51], [79] and increases the system complexity and computational load, thus compromising the system accuracy and robustness. Additionally, the processing of wide-band signal usually leads to high power consumption [80]. The high power consumption can be mitigated using a multiband approach in which the signal is split into sub-bands. The sub-band processing method will be briefly discussed in subsection III-C3.

Another advantage that creates a challenging situation is the short pulse nature of UWB signals. The coding of short width pulses requires longer synchronisation times, limiting the data capacity. Moreover, the short-width pulses increase the number of multipath components [77], which also compromises the overall system performance. Researchers have proposed a solution for this issue by devising special schemes and protocols to avoid repeated synchronisation [77]. In addition, the authors of [81] proposed the use of multipleinput multiple-output systems to mitigate the effect of short communications. One last disadvantage of UWB is its limited usage outdoors. According to various countries' regulations, fixed UWB transmitters operating outdoors are not allowed [82], [83], refer to subsection III-B for more details.

#### **B. UWB LICENSING AND REGULATIONS**

The UWB bandwidth license is free for indoor applications, yet the regulations for UWB devices are country- or regionspecific to define the technical requirements and certification procedures for legal and safe operation [84], and more importantly, to minimise the potential interference to licensed services [69]. These regulations comprise the boundaries and safety limits of the operating frequency, power levels, emissions, energy disruptions, service times and antenna locations. For example, the regulations for UWB devices in the United States are published by FCC in 2002 under the "Code of Federal Regulations Part 15, subpart F" [82], while those in the EU region are issued by the Harmonised European Standard in 2016 (the process started in 2006) under the radio equipment directive "ETSI EN 302 065 – 1 to 5" [83].

Additionally, these official UWB regulations distinguish between the different types and usages of UWB devices, and each type and usage has its own regulation. For example, the FCC has set specific rules for each category of UWB devices and their respective application, such as indoor UWB systems, handheld UWB devices, GPRs and wall imaging systems, surveillance and transportation systems (e.g. UWB on-board aircrafts and UWB installed on rail vehicles) [69].

According to the FCC, the bandwidth of the UWB systems belonging to the indoor and the handheld categories must be kept between 3100 MHz and 10,600 MHz [82]. The indoor UWB systems may not be used outdoors, and they must be designed so that they are capable of operating only indoors. The emissions from UWB devices may not be intentionally directed outside of a building to perform an outside function. Also, the use of outdoor mounted antennas is prohibited. The device may only transmit when sending information to an associated receiver.

An UWB device belonging to the handheld category must be relatively small. These devices are primarily kept in hand while being operated, and they do not employ a fixed infrastructure [82]. Antennas may be mounted only on the handheld UWB device. The use of antennas mounted on outdoor infrastructure is prohibited.

Part 1 of the EU regulation "ETSI EN 302 065" contains requirements for generic UWB applications, and it applies to fixed (indoor only), mobile or portable applications [83]. The UWB transmitter conforming to that document may not be installed at a fixed outdoor location, for use in flying models, aircraft and other forms of aviation. Allowed operation frequency band is from 3.1 to 4.8 GHz and from 6.0 to 9.0 GHz.

Requirements for UWB location tracking are defined in Part 2 of the EU regulation "ETSI EN 302 065". This document covers three types of UWB location tracking system, of which two are applicable for smart logistics applications [85]:

- LT1 systems: These systems, operating in the 6 GHz to 9 GHz region, are intended for general location tracking of people and objects. They operate on an unlicensed basis. The transmitting terminals in these systems are mobile (indoors or outdoors), or fixed (indoors only). Fixed outdoor LT1 transmitters are not permitted.
- LT2 systems: These systems, operating in the 3.1 GHz to 4.8 GHz region, are intended for person and object tracking and industrial applications at well-defined locations. The transmitting terminals in these systems may be located indoors or outdoors, and may be fixed or mobile. They operate at fixed sites and may be subject to registration and authorization.

The regulation documents contain additional points describing the operation peak powers and tabulated emission limits for UWB devices, which vary regionally (e.g. US and EU). Both the ETSI and the FCC regulations allow the use of UWB indoor location tracking, which is very important for many industrial and smart logistics applications. However, the unlicensed outdoor use of UWB is limited to handheld or mobile devices. Because the FCC or LT1 of ETSI do not allow fixed outdoor transmitters, development of UWB outdoor positioning systems becomes difficult. Without transmitting anchors, it is not possible to use TW-TOA and multilateration for position determination. In addition, TDOA scheme with wireless clock synchronization is inapplicable, since

the anchors must transmit synchronization messages to each other. Thus, the only possible way to implement outdoor UWB location system, operating under the provision of FCC or ETSI LT1, is to use TDOA approach with a wired clock synchronization. However, implementing the wired clock synchronization is complex and expensive.

LT2 of ETSI allows the fixed outdoor transmitters in the EU, but the LT2 systems are subject to registration and authorization. In addition, local coordination with possible interference victims has to be performed, and the possible permission would be granted only to a specific site [85]. Currently, developing an UWB-based positioning system for outdoor environment is very difficult.

In June 2020, the fine-ranging alliance (FiRa), the largest UWB consortium, was founded to pave the way for the widespread adoption of UWB-driven applications. Some well-known company members of the FiRa alliance are NXP, Samsung, Qorvo, Qualcomm, Cisco, Apple and BOSCH. The FiRa consortium is committed to providing seamless user experience through secured fine ranging and positioning capabilities of interoperable UWB technologies [86].

#### C. UWB SIGNAL ATTRIBUTES

The earliest attempt of UWB standardisation within IEEE standards was made by the WiMedia alliance workgroup in IEEE 802.15.3a-2003. This workgroup was responsible for standardising the physical and medium access control (MAC) layers of UWB indoor signals for wireless PANs. The detailed technical aspects of a UWB signal are described in the currently, active UWB standard (802.15.4z-2020), which was developed by the "LAN/MAN Standards Committee" of the IEEE Computer Society [87]. UWB signals can be generated using different techniques, the most popular of which is the impulse radio (IR) method. However, there are several other methods that can be adopted in UWB signals into the following six categories:

#### 1) IMPULSE RADIO ULTRA WIDEBAND (IR-UWB)

The IR-UWB modulates the baseband signal through short pulses (order of nanosecond duration each), which have a low duty cycle to transfer information. The frequency spectrum characteristics of IR-UWB can be controlled by varying the pulse shape, phase, amplitude and duration to formulate the spectrum envelope of the signal. IR-UWB can be carrierbased, which requires an external high-frequency sinusoidal carrier signal and a mixer, or carrier-less, which can operate without a local oscillator (LO) in the transceivers, only using the baseband signal. The IR-UWB is typically the most adopted system and is standardised in the IEEE 802.15.4z UWB standard.

#### 2) DIRECT SEQUENCE ULTRA WIDEBAND (DS-UWB)

Direct sequence spread spectrum (DSS) version of the IR-UWB forms the DS-UWB, which treats the signal by a pseudorandom number (PN) code before the amplitude

modulation of a train of short pulses. The new bandwidth of the transmitted signal is affected by a spread code, which is typically much higher than the symbol rate at which the chip interval is longer than the pulse width.

#### 3) MULTIBAND ULTRA WIDEBAND (MB-UWB)

The orthogonal frequency division multiplexing (OFDM) version of the IR-UWB can be considered to form the MB-UWB, in which the total bandwidth is divided into multiple frequency sub-bands (minimum 500 MHz each) to occupy the spectrum efficiently. The MB-OFDM approach utilizes the quadrature phase-shift keying (QPSK) modulation with 128 subcarriers and five-band groups containing two or three bands each (14 sub-bands in total). The MB-OFDM recently received approval from ISO/IEC and ETSI.

#### 4) FREQUENCY HOPPING ULTRA WIDEBAND (FH-UWB)

It is a non-conventional carrier-based method in which the transmission occurs through fixed frequency hops over a broad bandwidth and using variant frequency carriers. The hopping sequence is determined by a spreading code or a PN sequence set by the user in which a narrow-band transmission occurs periodically, which can be smaller than (fast hopping), greater than (slow hopping) or equal to the symbol rate. The total spectrum bandwidth is determined by the range of hopping frequencies and not the symbol rate.

## 5) STEPPED FREQUENCY HOPPING ULTRA WIDEBAND (SFH-UWB)

SFH-UWB is a particular case of FH-UWB, in which the hopping frequencies are selected by the spreading code to form linearly increasing discrete steps until the desired bandwidth is achieved. Then, the hopping frequency is reset to the starting sequence, and the process is repeated.

#### 6) SWEPT FREQUENCY ULTRA WIDEBAND (SF-UWB)

SF-UWB is also known as 'Chirp signalling'. It is the frequency variation of the FH-UWB, in which the carrier frequencies of the UWB waveform are generated by a voltagecontrolled oscillator using a continuous variable speed. The symbols are modulated on the slope (chirp) using M-ary modulation and then sent sequentially or superimposed.

#### D. ARCHITECTURE OF UWB POSITIONING SYSTEM

A typical UWB indoor positioning system includes fixed UWB sensors (anchors), mobile UWB targets (tags), location server and system interface. The location server stores and processes the sensors data, and the system interface (e.g. smartphone, computer or tablet) is for viewing the positioning results, as illustrated in Figure 4. Planar, two-dimensional positioning requires at least three anchors to solve the coordinate equations of the tag, while three-dimensional positioning requires at least four anchors.

Additional optional units can be added to the previous structure to obtain a real-time location system (RTLS). For example, the location server is optional in small-scale



FIGURE 4. Elements of UWB positioning system. User with a tag or mobile phone is located through UWB anchors. The location server can determine the position and provide an API for accessing the location information.

systems but crucial in large-scale systems. There are additional front-end and back-end units for complex indoor environments, such as navigation framework, network gateways, user interface and facilities for IoT integration or other accompanying multi-sensor technologies.

The process of UWB precise positioning commences with relative positioning between the anchors. A single initiator anchor is specified as a reference point or origin (0,0). An auto-positioning feature, such as Decawave's (Qorvo) RTLS application, measures the relative distance between all anchors and thus positions them in the coordinate system. A block diagram depicting an example of a complete process of UWB precise positioning is illustrated in Figure 5.

After fixing the coordinate system, the UWB system starts ranging the mobile UWB tag(s) within the indoor environment before sending the measured raw data to the positioning framework for additional processing. The positioning algorithm, which is pre-specified by the user, uses the raw measurements and a kinematic model to carry out position estimation. Precise position can be achieved by using the UWB system when the ranging method and the positioning algorithm are appropriate to the application and the properties of the environment.

Many applications in various environments require specific NLOS mitigation methods to improve the performance. This article focuses on two NLOS mitigation approaches, multi-sensor fusion and ML algorithms. Both approaches are discussed in detail in section VIII.

In the multi-sensor fusion approach, additional accompanying IPS technology is used to aid the UWB system with a fusion algorithm that fuses data obtained from all sensors based on their weights and shares.

In contrast, the ML approach is designed using large offline data to train a learning algorithm to identify the outlier measurements caused by the NLOS conditions. This approach has its performance metrics as ML algorithms are assessed from the training and testing accuracies. Nevertheless, the overall efficiency of the system is determined by the combined metrics of each phase, in addition to the degree of relevance of the final positioning results to the ground truth.

#### E. COMMERCIAL PRODUCTS

The growing demand for location-based services in an indoor environment has increased the size of the UWB market during recent years. In addition, the recent advances in UWB technology have provided opportunities for new commercial applications.

The major manufacturers providing UWB chips for open markets are Qorvo and NXP. Qorvo entered the UWB market by acquiring the Irish semiconductor company Decawave in January 2020 [88]. Decawave has been one of the major providers of UWB technology during the past 15 years, along with some other companies, such as Ubisense and BeSpoon [89]. Decawave's UWB technology has been very popular and its chips have been used extensively in research [16], [90]–[93] and commercial IPS [94]–[97].

NXP launched its UWB precision chips in February 2020. Currently, NXP provides Trimension UWB modules for IoT, industrial, mobile and automotive market segments [98]. Trimension UWB modules can be used, for example, like tags or anchors in IPS systems, in mobile devices and in secure car access applications.

One of the most important markets for UWB is mobile phones. In 2019, Apple launched iPhones having an UWB chip called U1 [99]. Samsung released its first high-end mobile phones with UWB technology in 2020. Samsung was one of the founders of the FiRa consortium together with NXP and some other companies [100]. UWB technology provides new opportunities for mobile phone use cases, such as secure access control, location-based services and deviceto-device communications. To support third-party application development, Apple has released its "nearby interaction" framework for developers and chipset manufacturers building UWB-based applications [101].

The automotive industry is developing UWB applications for secure access control and localisation. For example, Bosch's Perfectly Keyless key management system utilizes UWB in mobile phones for secure access control [102]. The vehicle access and start are controlled via a digital key on a mobile phone and the precise localisation of the phone.

#### **IV. UWB POSITIONING TECHNIQUES**

The positioning can be based on either multilateration or multiangulation techniques. When using multiangulation, the position of the unknown tag can be determined from known anchors geometrically by observing the angles of the received



FIGURE 5. Suggested building blocks of the UWB precise positioning process, based on the surveyed UWB literature.

signal either in anchors or in tags. When using multilateration, the ranges between the anchors and tags are obtained by measuring the time of flight (TOF) and multiplying it by the speed of light or by applying a channel attenuation model to the received signal strength (RSS) observations.

In general, the following four active strategies are used for estimating the TOF needed for multilateration:

- TOA: TOA of the signal is measured and subtracted from the known transmission time. The arrival timestamps are obtained by either using a precise clock (synchronous TOA) or by solving them together with the tag position by using at least N+1 anchors, where N is the number of spatial dimensions. The latter method is used in GNSS systems and is known as the pseudorange method. The clock synchronisation accuracy of the sub-nanosecond range is not feasible for moving targets [103], [104], and therefore, the pseudorange method is typically used.
- RTT: The RTT of signal propagation between two objects is measured, and the processing time is reduced to obtain double ToF. This method is also called two-way ranging (TWR) or two-way TOA (TW-TOA). The measurement is repeated between a tag and at least N anchors. This method is sometimes called asynchronous TOA. [105]
- Time of transmission (TOT): The arrival time of the signal sent by a tag is observed by at a minimum of N+1 anchors, and the TOT is solved together with the position of the tag.
- TDOA: The signal sent by the tag is received by at least N+1 anchors, which calculate the time differences of arrival times, avoiding the need for absolute time synchronisation in a tag.

The positioning techniques used with UWB are (1) TOA or TW-TOA; (2) AOA; (3) received signal strength (RSS); (4) TDOA and (5) a hybrid algorithm [10], [104]. The typical UWB system estimates position using a two-step procedure. In the first step, observables related to an unknown position of the tag and known positions of the anchors are determined. Among these, TOA, TDOA, AOA and RSS are given as examples [25], [26]. From these measurements, the RSS is considered the least suitable for use with UWB because its accuracy in an NLOS and a multipath environment is lower than that of time-based methods [104]. In the second step, the position of the tag can be computed using estimation methods such as least-squares (LS) method, Kalman filters (e.g. EKF and UKF) and particle filter (PF). In addition to the previous methods, some recent literature has adopted the AI approach through supervised learning to account for the position estimation in which the raw UWB data are compared with a trained ML model (e.g. support vector machine (SVM)) for predicting the unknown position of the target node [106]. Selecting a suitable positioning technique is vital to the whole precise positioning process as it affects the overall accuracy and defines the system complexity, and hence, the resources' total costs [76]. In this section, we summarise the most widely adopted techniques in the literature, highlighting their methodology, pros and cons.

#### A. ANGLE OF ARRIVAL

The position of an object can be estimated from the AOA or the angle of departure (AOD) of the signal. Each angle measurement defines a line between the base station and a mobile device. The object's location is determined from the intersection of these lines, as illustrated in Figure 6 a.



FIGURE 6. Ranging and angulation techniques. a) AOA, b) TOA, TWR and RSS and c) TDOA. In AoA, TOA and TDOA techniques, the tag position is determined from the intersection of the lines, circles and hyperbolas, respectively.

In the AOA method, the tag transmits a signal using a single antenna, and the anchor (base station) receives the signal with multiple antennas arranged in an array. The signal direction is determined from different propagation delays of the signal between multiple antennas of the receiver antenna array and the single transmitter antenna.

In the AOD method, there is a single antenna at the receiver and multiple antennas arranged in an array at the transmitter. Usually, the anchor (beacon) transmits the signal, and the tag receives it. The signal direction is determined from different propagation delays of the signal between multiple antennas of the transmitter antenna array and the single receiving antenna.

The advantage of the AOA (or AOD) observable is that there is no need to time-synchronise the anchor clocks. However, the antennas have to be precisely calibrated to the correct orientation. The AOA can be measured with various techniques, but currently, antenna arrays are mostly used in positioning systems.

There is a significant advantage of using a UWB signal over a narrow band signal in phase difference-based AOA estimation. Due to the short duration of the pulse, the UWB receiver can separate the direct signal from the reflected signal better than the receiver of the NB signal [92].

In UWB-based systems, the AOA measurements are often used together with TOA or TDOA measurements [92]. For instance, the authors of [107] proposed an AOA estimation method for a UWB positioning system using a lower-cost single-anchor system. A centimetre-level UWB positioning system was proposed by [108] using a mono-station TOA/AOA positioning method. In addition, the cell phone applications of Samsung and NXP use both the TOA and AOA observables [109]. The Ubisense positioning system uses both AOA and TDOA measurements, but according to [89], this system is less accurate than the Decawave system, which uses the TOA observable only.

Authors of [110] have developed a TDoA-Based positioning system using a single hotspot. This hotspot consists of anchors placed very close to each other. The tag to be localised is around the hotspot. The system estimates the

44424

range and AOA between the hotspot and the tag. According to the authors, the error in AOA estimate is less than 3 degrees when the target is in 15 m distance. The error in the estimated range may be 4 m.

#### **B. TIME OF ARRIVAL**

Most of the UWB-based positioning systems use the concept of TOA ranging to determine the user position. This concept is based on measuring the time taken for an RF signal transmitted by an emitter to reach a receiver. The time interval, due to signal propagation delay is multiplied by the speed of light to obtain the distance between the emitter and the receiver.

The TOA technique exploits trilateration to determine the position of the mobile users based on the range from the mobile unit to at least three (3D) anchors at known locations.

In the TOA method, the position is estimated by intersecting circles (2D) or spheres (3D) with radius  $r_i$  and centre  $(x_i, y_i, z_i)$ , as illustrated in Figure 6 b. The radius of the circle  $r_i$  is obtained from the propagation delay of the signal. Point  $(x_i, y_i, z_i)$  is the known location of the anchor.

The 3D location of the object  $(x_u, y_u, z_u)$  can be derived from the set of nonlinear equations as in Equation (2):

$$\rho_i = \sqrt{(x_i - x_u)^2 + (y_i - y_u)^2 + (z_i - z_u)^2}$$
(2)

where *i* ranges from 1 to 3 and references the base stations at known locations,  $(x_i, y_i, z_i)$  denote the *i*-th base station coordinates in three dimensions and  $r_i$  is the range measured from the *i*-th base station.

The TOA method illustrated above requires the anchors and tags to be accurately synchronised. To avoid the synchronisation requirement, the TWR method can be employed to measure the signal propagation delay. The range between two devices is determined through the two-way exchange of a message and by measuring its arrival time. This method is also known as two-way time-of-arrival (TWR-TOA) or RTT.

The simplest version of the two-way ranging cancels the effect of the clock offset between the terminals, but the clock drifts of the terminals can still cause significant error



**FIGURE 7.** Double-sided two-way ranging method to estimate signal flight time cancelling the effects of clock offset and drift.

in the signal flight time estimate. The error caused by the clock drift can be eliminated by making the two-way ranging measurement transaction two times [111]. This method called as double-sided two-way ranging is illustrated in Figure 7, and the TOF of the ranging message is expressed as in Equation (3):

$$\tau_f = \frac{1}{4} \left( \tau_{RTT1} - \tau_{d1} + \tau_{RTT2} - \tau_{d2} \right)$$
(3)

where  $\tau_{RTT1}$  is the RTT measured by the tag, and  $\tau_{RTT2}$  is that measured by the anchor. The terms  $\tau_{d1}$  and  $\tau_{d2}$  are the reply times of the anchor and the tag, respectively.  $\tau_{RTT1}$  and  $\tau_{d2}$  are measured using the tag oscillator, and both measurements are biased by the oscillator offset of the tag. Similarly,  $\tau_{RTT2}$  and  $\tau_{d1}$  are biased by the oscillator offset of the anchor. Doublesided TWR cancels these oscillator offsets.

#### C. TIME DIFFERENCE OF ARRIVAL

One method to obtain the signal propagation delay is the TDOA method, which measures the difference in the arrival times of two signals. The anchor clocks must be precisely synchronised, but the tags do not need to be synchronised. The tag position is obtained from the intersection of multiple hyperbolas (Figure 6 c).

The distance difference between the tag and the anchor where the signal arrives first, is expressed as in Equation (4): [112]

$$r_{i,1} = cd_{i,1} = r_i - r_1$$
  
=  $\sqrt{(x_i - x_u)^2 + (y_i - y_u)^2 + (z_i - z_u)^2}$   
 $- \sqrt{(x_1 - x_u)^2 + (y_1 - y_u)^2 + (z_1 - z_u)^2}$  (4)

where *c* is the speed of light,  $r_{i,l}$  is the distance difference between the first and the ith anchor,  $r_1$  is the distance between the first anchor and the tag and  $d_{i,l}$  is the measured TDOA between the first and the ith anchor. Equation (4) defines a set of nonlinear hyperbolic equations whose solution provides the 3D coordinates of the tag.

In principle, the TDoA approach can be implemented in two ways, called as unilateral and multilateral techniques [113], [114]. In the unilateral system, the anchors transmit a signal and the tags measure the TDOA from the received signals. Also, the anchors transmit a signal with different time delays to avoid a signal collision. The unilateral technique has some advantages, such as infinite scalability in the number of tags [115]. As in GNSS, in the unilateral TDOA system the location privacy is preserved, since the position is estimated in the tag. However, in the unilateral architecture the design of the tag is complex and it has high energy consumption [114].

Because of the drawbacks of unilateral approach, most of the UWB-based systems using the TDOA scheme are based on multilateral technique. In multilateral approach the tag sends one message, often called a blink, and the arrival time is then measured by multiple anchors with respect to a common time reference. These arrival times are then sent to a master node or server, which computes the TDOA estimates by subtracting the arrival time of the pivot anchor from those of the other anchors. Thus, the number of TDOA estimates is one less than the number of arrival time measurements.

Even though the multilateral approach is less scalable than unilateral approach, it can still support many more tags than the TW-TOA system. In the multilateral TDOA architecture a tag needs to send only one message per range measurement, while in the TW-TOA scheme, several messages are required [51]. Scheduling techniques are needed to avoid packet collisions in the TW-TOA approach, limiting the number of supported tags. In the TDOA approach, the tags need not to be aware of the anchors, which makes message scheduling easier than in TW-TOA. The authors of [51] investigated the scalability of UWB-based indoor positioning for TDOA and TW-TOA approaches with different MAC protocol combinations. In the mathematical model proposed by the authors, when using a TDOA approach and time division multiple access (TDMA), more than 6000 tags per second can be supported in a single domain shell. The drawback of the TDOA method is that the anchors have to be accurately synchronised. Because the radio signal propagates at the speed of light, 1 ns time offset in the anchor clock would introduce a 30 cm error in the estimated range. Decawave DWM1001 achieves measurement accuracy within 10 cm in the TOA mode, which is equal to 333 ps accuracy in the propagation delay measurement. The synchronisation accuracy must be even better to achieve similar accuracy in the TDOA mode.

The anchors of the UWB-based positioning system can be synchronised either through wires or wirelessly. In wired time synchronisation, the clocks of all anchors are synchronised by means of wires or fibres. In wireless time synchronisation, each anchor has its own clock running independently of the clocks of the other anchors. The time estimates of the different anchors are synchronised to the common time by sending synchronisation messages.

A time synchronisation mechanism for the TDOA-based UWB positioning system was investigated in [116]. The authors developed a test setup to measure the time synchronisation errors of wired and wireless approaches. The standard deviation of the wireless synchronisation was 400 ps, whereas that of the wired synchronisation was 133 ps. However, the time synchronisation of the anchors might be challenging. Wired clock synchronisation or more stable clocks makes the positioning system complex and expensive. Wireless clock synchronisation techniques lower the range estimation accuracy when unstable clocks are used. Many commercial UWB applications, such as Eliko [97] and Exafore [95], use the TW-TOA approach. Some UWB systems, such as Sewio [96] and Pozyx [94], support both the TDOA and TW-TOA schemes. According to Pozyx, the TW-TOA method provides more accurate position estimates, while TDOA is better suited for large-scale applications, which need multiple tags to be supported.

#### D. RECEIVED SIGNAL STRENGTH

The measurement of RSS is straightforward and is performed in most radio receivers. RSS decreases as the receiver-transmitter distance increases. This phenomenon can be used to estimate the location of a mobile device from the RSS measurements either by trilateration or location fingerprinting.

Radio signal attenuation is not only affected by the distance between the transmitter and receiver but also by multipath interference and any obstruction on the signal path. Thus, indoor positioning systems seldom compute the object position by using geometric range estimates derived from RSS. Instead, RSS-based indoor positioning systems use location fingerprinting more often. RSS is seldom used with UWB, because using the RSS observable does not completely exploit the benefit of UWB signals [117].

#### E. PASSIVE POSITIONING

In addition to previous active positioning techniques, location can also be determined by passively monitoring the communication of the UWB network [118] or by special arrangement; where a tag listens passively to the positioning messages sent by the anchors simultaneously [115]. Passive monitoring is not as accurate as active positioning, and the simultaneous positioning message method requires special hardware. Another passive UWB positioning strategy is to use UWB anchors as radars. In this case, the round trip time of the signal sent from the anchor and reflected from the target is measured [93], [119], [120]. Both magnitude and phase of the channel state can be used, in addition to signal reflections from walls. This method is called devicefree positioning since it does not require any UWB specific hardware in the target. Passive and device-free methods are still under research. They are not as precise as the active methods and are more sensitive to environmental changes and variation. Thus, they are not discussed further in this review.

#### **V. UWB POSITIONING ALGORITHMS**

The position can be estimated from TOA and TDOA observables by using various methods. When the position is estimated using measurements of a single time epoch, it is called static positioning, where the previous or future measurements are not accounted. The most common approach to solve the nonlinear system of equations of TOA and TDOA is to use the iterative LS method. Alternatively, the position can be estimated using closed-form solutions or methods based on likelihoods or probability.

Static positioning is not an optimal solution in most situations, as it does not account for the dynamic state model of the target. In many cases, a Kalman filter and PF provide a better position estimate, as they use the time series of the measurements for computing the current state estimate.

## A. ITERATIVE LEAST SQUARES AND CLOSED-FORM SOLUTIONS

The overdetermined and nonlinear system of equations can be solved using the Gauss-Newton algorithm. As the linearisation of this algorithm is based on Taylor-series expansion, it is also called the Taylor algorithm. In this algorithm, the user's position is determined using an iterative process starting from an approximate position.

The true distance between the anchor i and the tag is as described in Equation (5):

$$r_i = \sqrt{(x_i - x_u)^2 + (y_i - y_u)^2 + (z_i - z_u)^2}$$
(5)

where  $(x_i, y_i, z_i)$  is the position of the *i*-th anchor and  $(x_u, y_u, z_u)$  is the position of the tag. If  $(x_v, y_v, z_v)$  is the initial approximate position, let  $x_u = x_v + \delta_x$ ,  $y_u = y_v + \delta_y$ and  $z_u = z_v + \delta_z$ . By linearising Equation (5) using Taylorseries expansion and omitting the second-order and higher terms as in Equations (6)–(10) we have: [121]

$$H\delta = b \tag{6}$$

where

$$\mathbf{b} = \begin{bmatrix} r_1 - r_{v1} \\ r_2 - r_{v2} \\ r_3 - r_{v3} \end{bmatrix}, \quad \mathbf{H} = \begin{bmatrix} a_{x1} & a_{y1} & a_{z1} \\ a_{x2} & a_{y2} & a_{z2} \\ a_{x3} & a_{y3} & a_{z3} \end{bmatrix}, \quad (7)$$

$$\delta = [\delta x_u, \delta y_u, \delta z_u]^T$$
(8)

$$a_{xi} = \frac{x_i - x_v}{r_{vi}}, \quad a_{yi} = \frac{y_i - y_v}{r_{vi}}, \quad a_{zi} = \frac{z_i - z_v}{r_{vi}}$$
 (9)

and

$$r_{vi} = \sqrt{(x_i - x_v)^2 + (y_i - y_v)^2 + (z_i - z_v)^2}$$
(10)

The LS solution to the position estimation problem is obtained from Equation (11): [121]

$$\delta = (\mathbf{H}^{\mathrm{T}}\mathbf{H})^{-1}\mathbf{H}^{\mathrm{T}}\mathbf{b}$$
(11)

The position of the object  $(x_u, y_u, z_u)$  is calculated using an iterative process. In the beginning, the approximated position,

 $(x_v, y_v, z_v)$  is set to an initial value. Next, the direction cosine matrix **H** and the predicted-minus-observed range vector **b** are computed. Then, the unknown displacement vector  $\delta$  is calculated using Equation 8. The iteration process is repeated until the length of the displacement vector does not decrease any further.

The ordinary LS method described above assumes that the error variance in each measurement is the same. The method of weighted LS (WLS) can be used when the error variances of the measurements are not constant. WLS method weights observations by the reciprocal of the error variance  $w_i = 1/\sigma_i^2$  for that observation.

The weight matrix is calculated using Equation (12):

$$\mathbf{W} = \begin{bmatrix} \omega_1 & 0 & 0 & 0 \\ 0 & \omega_2 & 0 & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \dots & \omega_L \end{bmatrix}$$
(12)

The WLS solution is obtained from Equation (13):

$$\Delta \mathbf{x} = (\mathbf{H}^{\mathrm{T}} \mathbf{W} \mathbf{H})^{-1} \mathbf{H}^{\mathrm{T}} \mathbf{W} \Delta \mathbf{p}$$
(13)

In addition to the Taylor-series method, several closedform solution methods have been proposed to solve the set of TOA or TDOA equations. Caffery [122] presented a geometrical interpretation in which straight, rather than circular, lines of positions were used to determine the target device's position. This method is called the linear lines of position (LLOP) method [121], [123] or simply LS method [112]; it does not use linearisation. Another well-known closed-form solution is the Chan algorithm, which estimates the target position from TDOA equations [124].

Many authors have preferred closed-form solutions over the Taylor method [122], [124]. The Taylor-series method has been criticised as it converges towards a local minimum if the initial guess is not close enough to the true position. However, in a GNSS, for example, the Taylor-series method seldom converges towards the minima. The authors of [112] compared the LLOP, Chan algorithm and Taylorseries method and found that the Taylor algorithm provides the best positioning accuracy. A fusion algorithm combining both the Chan and Taylor algorithms can improve the positioning accuracy in the presence of NLOS errors [125].

#### **B. BAYESIAN FILTERS**

Recursive Bayesian state estimation, or Bayes filter is an abstract concept for tracking object's position in kinematic case, by combining a dynamic state model with observations. Bayesian filters recursively update the posterior belief to the current state as in Equation (14):

$$Bel(x_k) = p(x_k | y_{0.k-1}),$$
(14)

where  $x_k$  is the current state and  $y_k$  are the observations.

Practical implementations require the definition of dynamic and perceptual models and representation of beliefs. Depending on the implementation, the properties of Bayes filters are different [126]. Some most common implementations are different variations of Kalman Filters, Particle Filter and factor graph optimisation, which are described in the following subsections.

#### C. KALMAN FILTERS

The Kalman filter algorithm is a recursive estimation method used for predicting the new optimal states in linear state-space systems considering additive white Gaussian noise [127]. The algorithm is based on using a priori knowledge to estimate the posterior states, calculate the Kalman gain and measurement residual caused by the mismatch error and then calculate the new state and covariance vectors and use them as input to the next iteration [127]–[131].

#### 1) EXTENDED KALMAN FILTER

The extended Kalman filter (EKF) is an adapted version of the ordinary linear Kalman filter to estimate states in nonlinear dynamic systems [132]. A discrete-time Kalman filter follows two steps: 1) prediction step, where the next state of the system is predicted given the previous measurements fed to the system, and 2) update step, where the current state of the system is estimated given the measurement performed at the active time step [131], [133]. Then, the Kalman filter algorithm is used to satisfy the equations of state-space estimation in Equations (15) as follows:

$$x_{k} = f(x_{k-1}, k-1) + q_{k-1}$$
  

$$y_{k} = h(x_{k}, k) + r_{k}$$
(15)

whereas  $x_k$  and  $y_k$  are the state and measurement vectors of the system at time step k and  $q_{k-1}$  and  $r_k$  are the process and measurement noises at time step k-1, where  $q_{k-1} \sim N(0, Q_{k-1})$  and  $r_k \sim N(0, R_k), f(.)$  and h(.) are the nonlinear functions of model dynamics and measurement, respectively.

In EKF, the state transition matrix **F** and measurement matrix **H** in the linear Kalman filter are replaced by the nonlinear state transition function f(.) and nonlinear measurement function h(.), respectively, to map the algorithm through Gaussian distribution to work under nonlinear conditions. The complete Kalman algorithm for nonlinear systems is demonstrated in Table 3, which is adapted from [131].

Whereas  $m_{\bar{k}}$  and  $P_{\bar{k}}$  are the predicted mean and covariance of the state, respectively, at time step k before checking the measurement, and  $m_k$  and  $P_k$  are the estimated mean and covariance of the state at the time step k after checking the measurement.  $y_k$  is the measurements vector of the system at the time step k.  $S_k$  is the measurement prediction covariance at the time step k.  $K_k$  is the filter gain (i.e. the prediction correction coefficient at the time step k). f(.) and h(.) are the nonlinear functions of model dynamics and measurements.

#### 2) UNSCENTED KALMAN FILTER

Unlike EKF, the unscented Kalman filter (UKF) employs the sigma-point Gaussian transformation to map the nonlinear state transition function of the system and tends to

0 Initialization	for $k = 0$ set $\hat{X}_0, P_0^-, Q_0, R_0$					
1 Prediction step	Prior estimate of the state:	$m_{k}^{-} = f(m_{k-1}, k-1)$				
	Prior estimate of the covariance:	$P_{k}^{-} = F_{x}(m_{k-1}, k-1)P_{k-1}F_{x}^{T}(m_{k-1}, k-1) + Q_{k-1}$				
	Measurement residual update:	$V_k = y_k - h(m_k^-, k)$				
	Measurement covariance update:	$S_k = H_x(m_k^-, k) P_k^- H_x^T(m_k^-, k) + R_k$				
	Kalman gain calculation:	$K_k = P_k^- H_x^T(m_k^-, k) S_k^{-1}$				
2 Update step	Updating the posterior state:	$m_k = m_k^- + K_k V_k$				
	Updating the posterior covariance:	$P_k = P_k^ K_k S_k K_k^T$				
	Return to step 1, repeat until $k$ iterations are consumed.					
Output	Estimated state vector: $\hat{X}$					

TABLE 3.	EKF algorithm	for nonlinear	systems,	adapted from	[131].
----------	---------------	---------------	----------	--------------	--------

linearise it through the so-called unscented transform [131], [134], [135]. In other words, while the EKF approximation relies only on one point (the mean), UKF uses more than one point, including the distribution mean. UKF selects additional weighted points (called sigma points) plus the mean for more accurate transformation. This procedure is called the unscented transform. Thus, UKF sometimes outperforms EKF in severely nonlinear systems, whereas EKF performs well in systems with modest nonlinearity [136]. In an ideal case, both EKF and UKF can be used to solve the spatial positioning equation when Newtonian equations of motion are used to form the state transition function and the measurements from motion sensor (e.g. inertial unit, gyroscope or accelerometer) are being filtered. However, various recently proposed approaches use measurements obtained from various sensors to input as the state-transition matrix. The authors in [137]-[139] implemented a fusion positioning method, whereas the state-transition function originates from inertial navigation sensors, such as the inertial measurement unit (IMU) or inertial navigation system (INS), while the measurement function is obtained from UWB sensors. The two streams are fed to EKF/UKF for optimal positioning. The procedures vary from method to method. In some cases EKF is used with multiple anchor readings and a single-observation anchor, whereas another method involves the use of EKF and UKF as a cascaded system to obtain the input parameters of the second filter from the outputs of the first filter. The results showed that the fused UWB, along with the inertial sensor data exhibited improved overall positioning accuracy and system robustness.

Kalman filters also perform well when they coexist with PFs, as in [136], where the authors developed a framework comprising three filters (EKF, UKF and PF) with detailed evaluative metrics. The results showed that the developed structure could be used for numerous purposes besides positioning improvements, such as target tracking and robot localisation. Numerous realisations that involve the use of Kalman algorithms and their variations have been proposed in [22], [34], [51], [136], [140]–[151].

#### **D. PARTICLE FILTERS**

Particle Filter (PF) is another realization of Bayes filters for position estimation. PF is a popular choice for positioning because it can be used for solving the DSS shown in Equation (15) without assuming that the dynamic or perceptual models are linear and that the noise is Gaussian. However, it is also possible to implement a PF which assumes Gaussian posterior distribution if a lighter but more restricted version is needed [152]. The reasons for using PF are that the conditions of linearity and Gaussian noise do not hold very often, and linearisation is only possible if the model is well known [152], [153].

PF estimates the posterior belief  $Bel(x_k)$  of DSM through a sequential Monte Carlo (SMC) algorithm. The SMC is similar algorithm shown in Table (3), without resorting to linearisation nor Gaussian noise assumption. The posterior distribution can be anything representable by discrete samples (particles). Increasing the number of particles makes it possible to describe more complex distributions, but it also increases the computational cost of the method.

The algorithm is simple to implement, consisting of the following steps:

- 1) **Initialisation**: *N* particles are initialised according to the *a priori* knowledge described as probability distribution  $p(x_{k-1})$ . This distribution can be any suitable empirical distribution which can be presented with particles, or uniform distribution, if more informative distribution is not available.
- 2) Estimation loop
  - a) **Predict**: All particles are moved based on the current DSM by sampling new particles,  $x_k^i$ ,  $i \in [1, N]$ , from the distribution obtained by convolving the *a priori* distribution with the process model:  $\int p(x_k|x_{k-1}, u_k)p(x_{k-1})dx_{k-1}$ . The process model includes also possible inputs,  $u_k$ , and process noise.
  - b) **Update**: The weights of particles are updated according to the belief of the observations,  $w_k^i = p(y_k | x_k^i)$  assuming that the particle,  $x_k^i$ , represents

the correct location. Finally the weights are normalized so that their sum is 1.

- c) **Resample**: New set of particles is generated by sampling the existing set of particles using their weights so that particles with higher weights are selected more probably than the others.
- d) **Estimate**: The position estimate is obtained as a weighted average of the posterior distribution represented by particle locations and weights.

In NLOS conditions, the probability distribution of the position can be multimodal [154], and De Angelis *et al.* showed that ranging, which is based on RTT measurements can be affected by non-Gaussian noise even in LOS situations [143]. Therefore, PF can be more competitive in some realistic indoor environment than LS or EKF.

Many researchers have compared the performance of PFs with classical solutions for UWB-based indoor positioning. Usually, the performance is evaluated by examining the RMS error [155] or more often with the cumulative density function (CDF) of the positioning error in one to three dimensions [143], [148], [156]. These benchmarks show that, in many cases, a PF provides more accurate positioning results than classical methods, but in some cases, spurious errors are also detected when applying a PF (e.g. in the case of a kidnapped robot) [157]

It is also claimed that, instead of using the RMS error, comparing the whole posterior distributions of algorithms is more affective [152]. For example, the posterior distribution provided by a standard PF was assumed to be the most accurate and was compared with the Gaussian posterior distributions provided by EKF, UKF and GPF through Kullback-Leibler divergence and  $\chi^2$  information metrics. In this benchmark, the GPF was found to be more accurate in location tracking than EKF or UKF but incurred higher computational cost [152].

In addition to the increased computational cost, the problems identified in applying PF are sample degeneracy and impoverishment caused by the reduction in particle diversity [158]. A PF might also perform poorly in the kidnapped robot case when the robot is suddenly transferred to another location without allowing it to make measurements during the transfer. In this case, there might not be any particle near the actual position of the robot, and the robot might take a long time to find its new location. Counterintuitively, a PF does not also perform well when the measurement noise is too little [159], which is precisely the case in the controlled UWB positioning system. However, many methods have been proposed to overcome these problems. For example, dual MC localisation is a solution for too accurate sensor readings, and the kidnapped robot case can be solved by uniform particle augmentation [159]. Some other improvements are evolved distribution sampling methods [160], particle resetting approach [155], [161] or replacing PF with adapted FIR filter [157]. Zhu et al. improved the accuracy of a PF in UWB positioning by using a pre-build error distribution map [162].

At the cost of pre-computation, they gained increased 2D positioning accuracy.

It is relatively easy to fuse information from many sources into the PF estimations to compensate for NLOS issues, for example. Many researchers have fused INS sensors [91], [148], [156], mixing an EKF as a pre-processor of PF information [148]. The PF positioning algorithm can also include other models, such as UWB uncertainty model [91] or a model to predict the UWB signal obstruction caused by a pedestrian's own body [163]. The authors in [164] showed that the inclusion of digital maps into the PF model improved the positioning accuracy. The positioning of the anchor itself can be included in the PF-based positioning system [156].

#### E. FACTOR GRAPH OPTIMISATION

Factor Graph Optimisation (FGO) is a relatively new positioning algorithm among Bayesian filters. Some earlier publications from 2012 propose using FGO for multipath mitigation of GNSS positioning [165] and for multi-sensor fusion of GPS, IMU and stereo vision [166]. FGO models previous states as nodes and measurements as factors. Like KF and PF, FGO assumes Gaussian noise and utilizes the Bayesian filtering principle for solving the position estimation. The differences are that the FGO does not assume the Markov condition, but it uses information from previous states in addition to utilizing the latest state only. FGO solves the position by optimising the factor graph model with an iterative solver, therefore requiring more resources than KF or PF. However, it is still solvable in real-time, for example, by combining expectation-maximization and nonlinear optimisation methods [167], [168].

Even though FGO shares the unimodal Gaussian model with EKF, it can be more reliable in urban canyon environments for GNSS positioning cases [169]. Recently FGO has raised plenty of interest in positioning research, and it has been also applied to indoor navigation, including UWB positioning [167] and tight coupling of UWB and INS [170]. Besides, FGO can be useful in indoor positioning where multipath propagation causes channel impairments or in complex multi-sensor fusion situations.

#### F. PARTICLE SWARM OPTIMISATION

Particle Swarm Optimisation (PSO) algorithm belongs to the family of swarm intelligence, which also includes, for example, artificial Bee colony (ABC), Ant colony (AC) and Firefly algorithms. PSO was originally presented by Eberhart and Kennedy [171], [172]. PSO is an iterative global search technique that imitates the social behaviour of the swarm of birds. The algorithm is initialized with a random population of candidate solutions from a *D*-dimensional search space. Each candidate solution has a position and a velocity, and it is thus described as in Equation (16):

$$(x_i, v_i) = ([x_{i,1}, x_{i,2}, \dots, x_{i,D}]^T, [v_{i,1}, v_{i,2}, \dots, v_{i,D}]^T)$$
(16)

The quality of a solution represented by each particle is estimated by evaluating the loss function  $L(x_i)$ . The position of the best solution found by a particular particle is stored as the best local solution,  $P_{i,\text{best}}$ , and the position of the best solution among all particles is stored as global best  $G_{\text{best}}$ .

In each simulation step, the velocity and the position of each particle are updated according to the formula (17): [126]

$$\begin{cases} v_i^{k+1} = w v_i^k + c_1 \xi(P_{i,\text{best}} - x_i^k) + c_2 \eta(G_{\text{best}} - x_i^k) \\ x_i^{k+1} = x_i^k + r v_i^k \end{cases},$$
(17)

where w is the weight coefficient corresponding to the inertia of the particle,  $c_1$  and  $c_2$  are respectively the self cognition and social knowledge coefficients determining how much the model utilizes local knowledge vs. swarm knowledge. Stochasticity to the model is provided by selecting random variables  $\xi$  and  $\eta$  from range [0..1]. Position updating ratio is constrained by a constant factor r.

Some examples of using variations of PSO in positioning are: the use of ensembles of particle swarms to enhance the robustness and accuracy of UWB positioning [173], [174], and [175]. Being a global search strategy, PSO can be useful in finding the global minimum from the search space, but since the UWB positioning problem is often unimodal, simpler optimisation methods are usually more effective. However, in NLOS conditions, for example, the probability of the position can be multimodal [154], then, the global optimisation strategies can be superior in a similar way to PF.

#### G. NLOS IDENTIFICATION AND MITIGATION

In addition to using PF and PSO algorithms, the robustness of positioning in NLOS conditions can also be increased by specific NLOS identification and mitigation strategies.

Although UWB has several advantages in indoor positioning, it can still suffer from errors caused by NLOS conditions. In an NLOS condition, only the reflected signal is received, while the direct path signal is missing, as shown in Figure 8. The use of reflected signals can cause a significant bias to the position estimate. Various methods have been developed to reduce the inaccuracies imposed by an NLOS condition in UWB-based positioning. The authors in [176] categorised these methods as NLOS identification and NLOS error mitigation techniques. The signal can be identified as an NLOS signal, for example, by analysing the channel statistics parameters, such as mean delay, excess delay, amplitude and SNR. When the faulty observable is identified, it can be excluded from further analysis to improve the positioning accuracy.

Various methods can be used for mitigating NLOS errors, such as outlier detection, PFs, ML and weighted least squares (WLS). The authors in [177] proposed an NLOS identification technique based on multipath channel statistics, such as the kurtosis, the mean excess delay spread and the RMS delay spread. A likelihood ratio test was conducted for LOS and NLOS identification. Smaller weights were assigned to the measurements, which were likely biased. The authors analysed the WLS method, which deployed the likelihood



FIGURE 8. NLOS and LOS situations in UWB positioning system. The anchors in the left and right upper corners do not have a LOS connection to the tag, but the signal is received only through reflections, and therefore, the flight time is longer than expected, and the distance estimate fails. The anchors in the lower corners receive the LOS signal first.

functions obtained from the multipath components of the received signal. Recently, ML has been extensively applied for NLOS identification and error mitigation, which is presented in Section VIII-A.

A commercial solution was coined by Decawave (Qorvo) in [178] for systems that contain the DW1000 chips. This embedded resolution comprises the use of additional registers to assign a level of confidence to the received timestamps. Afterwards, it post-processes accumulators to allow the identification of a falsely detected first path, hence, identifying NLOS situations.

#### VI. RECENT ADVANCES IN UWB POSITIONING LITERATURE

The main contribution of this paper is that it summarises the most recent advances in UWB positioning literature that have spanned the previous five years; hence, we provide a compact summary, as shown in Tables 4–9. The tables categorise each article concerning the publication year, ranging methods used, applied algorithms, whether they use the fusion-based technique and field of application, as well as a summarised description explaining the rationale, methodology and findings for each article. In addition, several other articles are summarised as in-text review outside the table, which can be found in the relevant sections.

#### **VII. ADVANCES IN SENSOR FUSION TECHNIQUES**

Sensor fusion is a computational procedure to combine the measurements from multiple sources such that the output information after fusion is maximized [179].

Ref	Year	Ranging method	Algo- rithm	Multi- sensor	Application	More information
[192]	2020	TOA	LS trilatera- tion	-	General	Improved the UWB positioning accuracy in low SNR situations by proposing a modified leading edge detection algorithm in addition to LS trilateration filtering.
[140]	2020	TW- TOF	KF, MLE	-	2D & 3D posi- tioning	A comparative study employing various signal processing and AI techniques to identify measurement outliers. LOS/NLOS classifiers were investigated in both 2D and 3D localisation.
[173]	2018	TWR, TDOA	ELPSO	-	2D & position- ing, 3D	Developed an ensemble learning particle swarm optimisation (ELPSO) method for UWB indoor positioning. The system optimised the positions of 36 tags in which the final estimated values converged after 20 iterations, which outperformed other PSO methods.
[194]	2018	AOD, AOA, TOA, RSS	RiMAX	-	2D & 3D posi- tioning	Developed a novel positioning algorithm using an extension to the RiMAX algorithm and based on the geometrical properties of the propagation path of the UWB signal. The performances of numerous ranging techniques were compared and exploited to achieve channel sounding measurements. The proposed method achieved an overall accuracy of 0.26 m in LOS and 0.90 m in NLOS.
[194]	2019	TOF	TSML- WLS	Wi-Fi	2D & 3D posi- tioning	Investigated the performance of Wi-Fi as an IPS when it cooperated with UWB-IPS. Experimental results showed increased accuracy when Wi-Fi access points were gradually replaced with UWB beacons (20- 30 cm per replacement). The proposed hybrid approach achieved an efficient compromise between positioning accuracy and infrastructure costs.
[141]	2020	TW- TOF	KF, MLE	-	2D & 3D posi- tioning	Proposed a real-time indoor positioning system for smart grids by employing UWB as IPS. The proposed framework is based on AI techniques to detect the measurement outliers caused by NLOS conditions and hence improve the positioning accuracy of smart grid components.
[195]	2020	TOA	Trilateratio LSTM	on <del>,</del>	2D positioning	Achieved high localisation accuracy (MAE = $0.7 \text{ m}$ ) by using two LSTM algorithm networks to mitigate the effect of NLOS measurements.
[191]	2019	ΤΟΑ	PUFIR	INS	2D positioning	Introduced the problem of missing UWB measurements and proposed a fusion with INS augmented by a predictive unbiased finite impulse response (PUFIR) filter. The authors compared its performance with three other filters: KF, an ordinary UFIR and a predictive Kalman filter (PKF). The results showed that the PUFIR filter yielded less RMSE (0.5 m), more robustness than KF and a reliable navigation accuracy (maximum error = $2.28$ m) amid missing UWB range measurements.
[189]	2018	TOA	EFIR	INS	2D positioning	Proposed a federated EFIR algorithm to fuse UWB and INS measure- ments. The real test results showed that the EFIR filter outperformed the conventional federated EKF.
[196]	2020	TOA, TDOA	MLP, CNN	-	2D positioning	Proposed a transfer learning-based UWB NLOS identification scheme for an unmeasured environment. The proposed hybrid method showed better NLOS identification accuracy (98.33 %) at 48 times faster training time than conventional ML algorithms MLP and CNN.
[142]	2020	ΤΟΑ	SVD- FDCKF	IMU	2D positioning	Proposed an IMU/UWB multi-sensor system using the singular value decomposition federated derivative cubature Kalman filter (SVD)-FDCKF, which continuously corrects the IMU through UWB positioning observables to avoid the accumulation of IMU drift errors in a short time.
[143]	2016	ТОА	LS, EKF, PF	-	2D positioning	Simulated the distance measurements using RTT and compared the estimation performance of PF against those of LS and EKF. Results showed that the PF output is more reliable than other algorithms but on account of computational complexity.
[197]	2020	TDOA	WKNN- LSTM	-	2D positioning	The LSTM model was used to predict UWB measurements, which were later used to fix the actual ones. The corrected UWB measurements were fed to the WKNN model, and hence, the localisation was obtained. This method resulted in high precision accuracy of 20 cm.
[125]	2020	TOA	Chan- Taylor	-	3D positioning	The research focuses on mitigating the effect of NLOS error on indoor 3D positioning by achieving RMSE error of 0.2 m.

#### TABLE 4. Summary of UWB literature: General methodology development.

Ref	Year	Ranging method	Algo- rithm	Multi- sensor	Application	More information
[75]	2019	TOA	LSE- Taylor	-	Multi-users in dense environments	Developed waveform division multiple access (WDMA)-based UWB positioning for dense multipath and NLOS environments. The proposed (WUB-IP) method was complemented by a transfer learning approach (SHLA) to mitigate NLOS. The WUP-IP method showed a high precision positioning scheme (RMSE $< 2$ cm) compared to other methods (RMSE $> 10$ cm).
[198]	2018	TOA	WLS, IVM	-	NLOS identifica- tion	Proposed a novel ML NLOS identification method using WLS and IVM algorithms. The tests showed that IVM had better positioning accuracy (in terms of RMSE and CDF) than RVM and SVM.
[199]	2019	TW- TOF	ECTSRLS	-	NLOS mitigation	Developed an equality constrained Taylor series robust least squares (ECTSRLS) technique followed by a fuzzy comprehensive evaluation (FCE) to mitigate the effects of NLOS situations. The FCE-ECTSRLS method is used for channel identification to select the optimal set of ranges for the position estimation. The experimental results showed that the proposed method outperformed other algorithms by RMSE = $0.602$ to $1.063$ m.
[200]	2021	AltDS- TWR	LS trilatera- tion	-	Fingerprinting positioning	Presented the Fingerprinting-Assisted UWB-based localisation (FAUL) method to improve the accuracy of UWB localisation in complex indoor environments. FAUL combines fingerprinting and weighted trilateration techniques to reduce the localisation error (accuracy > 95%) when LOS situations are minimal.
[201]	2020	TW- TOF	CNN	-	Fingerprinting positioning	Utilized an ML model (CNN) to improve the accuracy of fingerprinting measurements. The data were divided as 70% for training and 30% for testing, and the CNN method outperformed SVM and random forest algorithms

#### TABLE 5. Summary of UWB literature: NLOS mitigation and integration with fingerprinting.

#### TABLE 6. Summary of UWB literature: IoT.

Ref	Year	Ranging method	Algo- rithm	Multi- sensor	Application	More information
[202]	2018	ΤΟΑ	LS, CI, TSML	-	ЮТ	Derived a statistical model based on an LS algorithm to improve the performance of the estimated pairwise distances before forwarding to the localisation algorithms. The method was validated against two algorithms: CI and TSML. The results showed a significant improvement in performance within scenarios of varying geometry, such as IoT scenarios.
[203]	2018	ΤΟΑ	-	RFID	Industrial + IoT	Highlighted the numerous attempts made to utilize UWB in various environments, including the outer space. Experiments carried out by [208] showed that the UWB/RFID method could achieve up to 4 cm accuracy.
[205]	2017	ΤΟΑ	Tri- lateration	RFID	IoT, energy har- vesting	Highlighted the characteristics of energy-autonomous tags that can achieve centimetre-level positioning accuracy. The authors demon- strated the UWB-RFID backscatter method, a promising candidate for precise positioning.
[79]	2018	TDOA, TWR	-	-	Tracking in dense environments	Investigated the capability of UWB Decawave DW1000 chip to support numerous tags simultaneously in dense environments. The authors concluded that the proposed method (TDOA-TDMA) can support up to 6171 tags (updates per second), outperforming other combinations of methods.

As sensor technology becomes more sophisticated (and owing to its erroneous nature), multi-sensor fusion has been trending recently. The reliance on multiple measurement devices in positioning applications can result in fewer uncertainties, and greater reliability and accuracy than depending on a single measurement sensor [180]. Numerous tracking systems can be fused with a UWB system to produce more accurate and reliable estimations. Common examples of these systems are GNSSs, inertial navigation systems (INSs), dead reckoning (DR), visual map matching (VMM) and computer vision. The optimal positioning estimations that result from fusing multiple positioning methods follow a unified framework, which is illustrated in Figure 9.

GNSSs (e.g. GPS, GLONASS and Galileo) provide satellite-based positioning estimations for outdoor environments within an acceptable error range. However, the signal suffers from multiple degradation factors, such as multipath fading, path loss and shadowing, which reduce its applicability in indoor positioning applications [147], [181].

INSs are highly reliable positioning systems, as they are not influenced by external factors. However, they accumulate significant errors over time [182], [183]. The main role of

#### TABLE 7. Summary of UWB literature: Autonomous systems.

Ref	Year	Ranging method	Algo- rithm	Multi- sensor	Application	More information
[144]	2020	DS- TWR	EKF	DR	AGV control	Proposed a fusion-based UWB/DR localisation technique to achieve 10 cm tracking accuracy for AGVs in harsh industrial environments.
[206]	2020	TOA, TDOA	GD, LS	-	AGV	Developed a low-cost UWB localisation system that provided precise positioning for AGVs using gradient descent and LS algorithms. The proposed method was validated against RTK data, and the result was a robust system capable of localisation both indoors and outdoors.
[183]	2019	ΤΟΑ	LS, SHFAF	INS	Autonomous robots	Proposed an INS/UWB/SHFAF method to deal with the time-varying noise, which accounts for the positioning errors of indoor autonomous robots. Through simulations and experiments, the proposed method achieved high positioning performance in terms of accuracy and robustness in dense environments.
[186]	2017	TOA	IAKF, AR	INS	Mobile robots	Tested an INS/UWB sensor fusion approach to tackle the problem of accumulated errors for tracking indoor mobile robots in real-time. A 2D kinematic model, improved adaptive Kalman filter (IAKF) algorithm and AR algorithm were used to identify the estimated position outliers. The results showed that IAKF outperformed the KF algorithm with an improved error of 0.24 m.
[136]	2019	ΤΟΑ	EKF, UKF, PF	-	Robot localisation	Presented and compared the performances of EKF, UKF and PF in positioning mobile robots. The simulation results showed that the PF performed fastest in the low-speed mode (1 m/s), while UKF performed fastest at higher speeds (2-10 m/s).
[147]	2020	TDOA	FKF	GNSS, DR, VMM	Intelligent trans- portation	Designed the fusion positioning algorithm of GNSS/UWB/DR with a federated Kalman filter (FKF). The results obtained from both the simulation and real vehicle testing showed that the proposed intelligent vehicle localisation accuracy was improved (MAE < $0.88$ m). The positioning accuracy could be improved, and an adaptive information distribution coefficient was established based on the FKF.
[207]	2021	TOA	MLKF	-	UAV 3D posi- tioning	Indoor localisation of a fleet of vehicles using UWB. The mea- surements involved the calculation of inter-vehicle distance, which improved the overall positioning accuracy by 60 percent.
[149]	2018	DS- TWR	EKF	IMU	UAV 3D posi- tioning	Proposed a 3D attitude and pose estimation method for small UAVs involving the use of IMU, UWB and EKF filtering. The best RMSE values achieved for attitude and pose were 1.93° and 0.19 m, respectively.
[208]	2017	TW- TOF	LMA multilat- eration	-	UAV tracking	Developed a numerical method which involved the use of UWB technology and Levenberg-Marquardt algorithm (LMA) to track and control UAVs. The analysis results showed that an improved accuracy on decimetre order could be achieved for the 3D-positioning of UAVs.
[32]	2020	TOA, TDOA	Cost function	GNSS, LoRa	UAV traffic man- agement	Proposed a high-precision prototype for a UAV traffic management (UTM) system. Besides GPS, the authors used a novel method comprising an IR-calibrated UWB (Decawave) in addition to the LoRa module. The proposed cost function reduced the GPS positioning error from 4.03 to 1.73 cm.



FIGURE 9. Fusion-based positioning framework, adapted from [179], [180].

UWB precise positioning technology is to refine INS errors by tightening the position estimate to the absolute coordinate system, while an INS provides more accurate delta position updates in the short term, making the integrated INS/UWB system more accurate and robust [26], [76]. An IMU differs from an INS as it is not an integrated dynamic system as an INS. However, IMU units are the main building block of an INS [184]. IMUs can still be used independently in fusion-based localisation endeavours, but INSs have recently been widely adopted in positioning systems.

## A. SENSOR FUSION POSITIONING IN TRANSPORTATION APPLICATIONS

As cooperative positioning is a crucial element in intelligent transportation systems (ITSs), the authors in [185] developed a cooperative scheme supported by vehicle-to-infrastructure (V2I) communications as a prototype implementation of an

#### TABLE 8. Summary of UWB literature: Industrial applications.

De	¥7.	n	41	N/ 14	A . 1'	
Ref	Year	Ranging method	Algo- rithm	Multi- sensor	Application	More information
[34]	2020	TW- TOF	EKF, ESKF	IMU	Coal mine robots	Developed an IMU/UWB-based localisation system to equip CMRs with reliable estimations that improved navigation in underground tunnels. UWB measurements were filtered by EKF, which were federated with IMU measurements through the ESKF algorithm. Small-scale experiments showed improved robustness but less positioning accuracy (RMSE = $0.562$ m).
[191]	2019	TDOA, AOA	KF	SINS	Coal mines	Proposed the SINS/UWB method, in which both measurements were fed to a Kalman filter to fuse the estimated position and speed. The results showed that the tightly coupled model and the decision tree model could produce accurate positioning during partial and total node failure situations. The maximum error of the SINS/UWB multimodel method was 0.93 m, accumulated in 327 s.
[149]	2020	ADS- TWR- MTMA	WLM, WLS, KF	IMU	Underground Mines	A numerical analysis using simulation data to validate the impact of using fusion-based method (UWB/IMU) to improve the localisation accuracy in underground Mines.
[209]	2020	ΤΟΑ	Chan	-	Substation safety monitoring	Investigated the feasibility of a test environment that employs UWB positioning in 500 kV substation operation monitoring. Despite the high costs and shielding problem, the experimental results showed positioning accuracy of 0.15-0.3 m in the UWB substation monitoring environment.
[90]	2017	AOA, TDOA	PF	-	Industrial warehouse	Conducted a comparative study between commercial UWB providers such as BeSpoon, Ubisense and Decawave. The authors concluded that the Decawave system outperformed other systems in terms of reliability and accuracy owing to its advanced antennas.

#### TABLE 9. Summary of UWB literature: Pedestrian positioning.

Ref	Year	Ranging method	Algo- rithm	Multi- sensor	Application	More information
[145]	2019	TW- TOF	EKF, K- means	-	Body wearable sensors	Investigated the effect of UWB wearable location with respect to the body on the overall positioning accuracy. The method involved the use of smoothing, filtering and clustering. The experimental results showed that placing the UWB wearable tag on the human chest produced the largest positioning error due to the highest NLOS component, while placing it on the forehead yielded the best accuracy.
[146]	2017	ΤΟΑ	EKF, EFIR	-	Human localisa- tion	Compared the performances of EKF and EFIR estimators for human localisation based on UWB. The results showed that EFIR performed better than LS and EKF algorithms, with MAE values of 0.19 and 0.54 m for the north and east directions, respectively.
[210]	2017	ΤΟΑ	PF, ZUPT	IMU	Pedestrian posi- tioning	Proposed a fusion-based method, IMU/UWB, through PF and ZUPT algorithms to achieve precise positioning for pedestrians. The results showed that the proposed method demonstrated high precision under NLOS conditions using fewer particles.
[148]	2020	TOA, TDOA	PF, EKF, ZUPT	INS	Pedestrian track- ing	The UWB observables were treated by joint particle filtering, while the INS data were filtered by the ZUPT algorithm. Both UWB and INS errors were loosely fused by EKF, resulting in improved accuracy (MAE = $0.35$ m) and better adaptability in robust environments.
[211]	2020	TDOA	SWA	-	Shopping mall	The TDOA ranging technique and sliding window algorithm (SWA) were used to track a moving stroller in a shopping mall down to 20 cm RMSE positioning error.
[51]	2018	ΤΟΑ	PF, EKF	-	Sports athletes tracking	Investigated the performance of UWB tracking for athletic activities, such as running, walking with varying speeds and accelerations and jumping. The tests involved PF and EKF algorithms, which showed an acceptable accuracy of approximately 20 cm.
[212]	2019	SDS- TWR	-	-	Track indoor cy- clists	Studied the feasibility of employing UWB for tracking indoor athletes (cyclists). The authors investigated various configuration setups to conclude the best positioning accuracy, best anchor height in LOS, least energy consumption and the most convenient spot for athletes to mount the mobile UWB tag. The experimental results showed that UWB positioning is viable for tracking indoor cyclists.

ITS. This method comprises a tightly coupled sensor-fused GPS/UWB/INS algorithm built upon an error detection unit,

in addition to a robust Kalman filter. The UWB component acts as a refining agent for noisy GPS measurements. The results showed that the fusion of UWB with GPS/INS improved the overall positioning accuracy of the vehicles down to a sub-metre level with an added pseudo-range gross error in different scenarios (the highest positioning error was 0.78 m) and, thus, increased the system reliability.

In ITSs, various low-cost positioning systems can be fused to improve the overall accuracy and reliability, as demonstrated in [147]. The authors used a federated Kalman filter (FKF) to combine GNSS, UWB, DR and visual map matching (VMM) in one framework, which the authors called an 'intelligent positioning strategy'. VMM is a common method in GNSS positioning, which uses pre-stored maps to correct the fusion error. Three input sources to VMM were propagated: (i) result of GNSS/UWB/DR fusion, (ii) captured images from a vehicle camera and (iii) pre-stored visual map repository, sampled by frame and pose. The integrated GNSS/UWB/DR/VMM strategy was tested in a simulation environment and a real-vehicle test platform on a university campus. The results of both tests showed that the framework improved the accuracy (MAE 0.9 m) and system reliability.

Intelligent logistics (also known as smart logistics) have recently adopted AGV robots which possess key transportation capabilities to maximise the efficiency of logistics traffic. The authors in [139] developed an INS/UWB integrated approach along with an interactive multiple model (IMM) algorithm that involves dual Kalman filters in both LOS and NLOS situations by combining their probabilities through a Markov chain transform. The INS and UWB position errors were fused, and the error covariance was updated using another Kalman filter, which compared the INS measurements against the UWB estimated values. Finally, the Kalman filter yielded the estimated position as the output and proceeded to the weighted fusion step. The results showed that the proposed INS/UWB-IMM method reduced the influence of a LOS/NLOS mixed situation; the average localisation error obtained was 0.2 m.

#### B. FUSION-BASED POSITIONING IN INDOOR APPLICATIONS

Fusion-based indoor positioning has recently gained significant attention due to advances made in wireless sensor networks, enhancements achieved in positioning technologies and its optimisation capabilities [180]. Another tightly coupled technique was presented by [186], where the authors used an INS/UWB sensor fusion approach to address the problem of accumulated errors in inertial navigation systems, which can localise and track indoor mobile robots in real-time. A 2D kinematic model of a mobile robot was developed for positioning and tracking, in addition to an autoregressive algorithm, to accommodate a third-order error equation for the gyroscope and accelerometer. In addition, an IAKF algorithm was developed, and a covariance matching method was used to identify the estimated position outliers. The results showed that IAKF outperformed the KF algorithm, and the error of the INS/UWB integrated



FIGURE 10. Structure of the federated Kalman filters.

system was improved to 0.24 m, which is an acceptable level for the practical requirements of the system model.

The concept of an FKF filter in multi-sensor data fusion has been implemented as a group of sub-filters corresponding to each sensor measurement in addition to the master combining filter, which regularly fuses sensor data to achieve optimal estimations [187], [188]. The concept of federal filters is illustrated in Figure 10. The authors of [189] adopted a federated extended finite impulse response (EFIR) filter as the sub-filter to fuse INS/UWB measurements between the reference nodes and target tag. Another EFIR was used as the master filter to achieve optimal position estimation based on the sub-filter outputs to mitigate the INS error.

The results obtained from [189] showed that the EFIR sub-filter approach was more accurate (RMSE = 0.45 m) and robust than the normal FEKF. In [190], the same authors introduced the problem of missing UWB measurements and proposed the combination of an INS augmented by a predictive unbiased finite impulse response (PUFIR) filter. The performance of the proposed method was compared with those of the other three filters: the Kalman filter, an ordinary UFIR and a PKF. Although the ordinary UFIR filter performed the worst, the PUFIR filter yielded a smaller RMSE (0.5 m) and more robustness than the Kalman filter and yielded reliable navigation accuracy (maximum error = 2.28 m) amid temporary missing UWB range measurements.

The authors of [183] addressed the multipath effect during NLOS situations on a UWB signal in dense, complicated environments for the indoor navigation of autonomous robots. The authors introduced an adaptive filter called Sage-Husa fuzzy adaptive filter (SHFAF) for outlier detection. Such filters assume that the noise is time-varying, especially in an NLOS situation, in contrast to Kalman filters, which assume time-invariant noise. In a SHFAF, noise covariance is estimated by adjusting the innovation weight adaptively, which results in more accurate estimations (88.2% of the time, the positioning error is less than 0.2 m) and enhanced robustness, as demonstrated by the simulation and experimental results.

## C. FUSION-BASED POSITIONING IN EXTREME CONDITIONS

GNSS signal is not available in underground mines, and all positioning systems in those conditions are prone to signal deterioration and multipath effects. Being quite tolerant against multipath propagation, UWB positioning has been used in Coal mine robots (CMRs) for precise underground positioning to function in excavation, mining and security control rescue tasks. The authors of [34] developed an IMU/UWB-based localisation system to equip CMRs with reliable Estimations, which can mitigate the navigation uncertainties in underground tunnels. The UWB measurements were propagated into an EKF. Subsequently, its output was federated with IMU measurements through an ESKF to realise six-degree-of-freedom state estimations. The estimates were compared with LiDAR odometry methods. The simulations and small-scale experiments exhibited improved robustness but slightly decreased positioning accuracy for the proposed fusion method ESKF-UWB over EKF-UWB. The authors attributed these results to two reasons: (i) the EKF method did not comprise orientation errors, or (ii) the crawler robot suffered movement vibrations, generating additional IMU noise. This implementation is yet to be investigated on a larger scale.

The strap-down inertial navigation system (SINS) is commonly employed in Chinese coal mines to measure the position and attitude of a shearer on rails; however, it suffers from the accumulation of error drifts over time. Hence, the authors of [191] proposed an integrated SINS/UWB system through a multimodel intelligent fault-tolerant algorithm to refine the positioning errors by switching between a tightly coupled model and a decision tree model based on the working state of UWB anchors. The switching was performed by assessing the ability of all anchor nodes to accurately measure the range between each stationary node and the mobile node. Afterwards, the determined UWB epochs, along with SINS estimations were fed to a Kalman filter to obtain the final position and speed. The results showed that the tightly coupled model could accurately localise the shearer amid partial node failure. In contrast, the decision tree model could produce accurate positioning during total node failure situations. The maximum SINS/UWB multimodel method error was 0.93 m, accumulated in 327 s.

#### VIII. ADVANCES IN MACHINE LEARNING APPROACHES IN PRECISION POSITIONING

The multilateration techniques directly estimate the position of an agent using distance observables with a direct signal propagation model. The reliability of the estimation can be increased using a dynamic state model, which in addition to current observations, uses the previous positions in estimations as well through Bayesian statistics (e.g. Kalman- and PF-based solutions). These are the most common methods and are particularly functional inside a controlled positioning infrastructure. While plain multilateration and dynamic state models have been the most common methods for positioning estimation, more advanced ML models have become increasingly popular due to their increased calculation capacity and advances in ML methods. They are used for positioning in situations in which simple models do not work efficiently (e.g. due to heavy nonlinearities, abruptly changing conditions, heterogeneous information sources, skewed noise distribution or non-convexity). Typical reasons for using ML are to increase robustness, allow adaptation to changes, implement a collaborative or model-free positioning system, use heterogeneous information sources or select the most useful features for positioning. These methods allow the use of ad hoc observations not originally intended for positioning, such as optical images and RSSs from Wi-Fi base stations. Often traditional positioning algorithms are supplemented with ML methods.

ML methods have been found helpful in many use cases, some of which are listed in the following subsections. Moreover, a summary that describes the properties, strategies and purposes of the reviewed ML algorithms is presented in Table 10.

#### A. RESOLVING NLOS UNCERTAINTY

UWB systems that are augmented by ML approaches play an important role in resolving NLOS situations. For instance, the basic ML algorithms, such as a *naïve* Bayesian filter and gradient descent algorithms were adopted by [213] and [206], respectively, to address NLOS conditions by recognising measurement outliers, hence improving the overall accuracy. Using large datasets, the authors of [106] proposed a radar system augmented by a multiclass support vector machine algorithm to localise and identify targets by specifying the location within the building rooms, which reduced the uncertainty associated with NLOS. A similar approach with a novel NLOS identification algorithm based on an import vector machine (IVM) algorithm, along with a feature selection strategy was proposed by [198].

The suitable performance of neural network deep-learning approaches have also been rising in dominance in the most recent UWB literature (in a span of the past three years). In [196], the use of a multilayer perceptron, with transfer learning and convolutional neural networks (CNNs) as NLOS classifiers, not only enhanced the overall training accuracy from 44% to 98% but also achieved faster training times than those achieved using CNNs alone in an unmeasured environment. Another neural network approach was presented in [195], where the authors employed a long short-term memory (LSTM) algorithm for predicting the user position based on the received TOA measurements. This LSTM approach resulted in a 7 cm accuracy, which outperformed several other techniques, including the recurrent neural network (RNN) method. The ML methods can be used to increase the robustness of the well-known model-based traditional methods. such as Kalman filter [214].

Algorithm	Purpose	Strategy	Properties	Ref.
Naïve Bayesian filter	NLOS mitigation	Outlier detection	Accuracy in terms of the area under curve (AUC) is $87\%$	[213]
LS + thresholding	NLOS mitigation	Outlier detection	Penalize the quadratic error (cost function)	[206]
CNN	NLOS mitigation	Outlier detection	Transfer learning, 98% accuracy	[194]
SVM	UWB radar	Venue detection	Correction rate > 95%	[106]
LSTM	Robust position- ing	Analysing mean error and hyper-parameters	Mean error 6-7 m, less than with other methods tested	[195]
Multi PF	Adaptive positioning	Change channel model when NLOS is detected	RMS positioning error 20 m in 2000 m field	[215]
Reinforced learning + PF	Adaptive positioning	Change PF mode accord- ing to success / failure	Accurate and fast recovery for positioning errors	[216]
PF + ML based data fu- sion	Collaborative po- sitioning	Peer to peer data fusion	Works during main positioning system outage	[217], [218]
PF + ML autocalibration	Collaborative po- sitioning	Automatic configuration	Autocalibration, adaptive time slot protocol	[220], [221]
Ensemble regression and deep learning	Model-free posi- tioning	Fingerprinting	Based on scanning a map of observables	[224], [225]

TABLE 10. Summary of machine learning methods used for positioning.

#### **B. ADAPTIVE POSITIONING**

The parameters of the prediction models can be tuned or changed according to the change in the operating conditions detected by ML models, for example, when an agent moves from outdoors to indoors or the detection of NLOS situations. Some researchers have been able to improve the positioning accuracy by using various dynamic-state models simultaneously and selecting the most suitable model according to the conditions detected by the ML model [215]. Neural networks and reinforced learning methods allow even more flexible adaptation to the changing environment [216].

#### C. COLLABORATIVE POSITIONING

Collaborative positioning means that agents share information with each other while performing positioning. The solution does not necessarily conform to the preconditions of the static or dynamic state model-based methods, as the noise distribution might be skewed, and the problem might be nonconvex. Therefore, traditional optimisation methods might not find the global optimum.

Hoang *et.al.* studied collaborative positioning of vehicles using onboard GNSS, IMU, odometer and UWB for inter-vehicle distance measurement. They formed an ad hoc communication network using the ITS-G5 standard for collaborative positioning [217]. The data fusion was performed using Bayesian frameworks with EKF and PF for pre-processing and data fusion, and they noticed that the PF outperformed the EKF. The experimental result showed that collaborative positioning using vehicle-to-vehicle communication could secure accurate positioning during GNSS outage in some cases [217].

The noise in DGNSS observables is not Gaussian, and the collaborative positioning problem is not convex. Therefore, the ML approach was used to create a cognitive PF for positioning for industrial IoT purposes [218]. They also noticed that if the noise probability distribution of the sensor data can be properly estimated and used in particle weight computation, the positioning accuracy can be further improved.

Infrastructure-free multi-robot localisation using UWB and PF [219]–[221], including functionality for system autocalibration [105], has been extensively studied. Both relative position and orientation of robots can be obtained by attaching several UWB requesters and responders in robots and measuring ranges using TW-TOA [222].

#### D. MODEL-FREE POSITIONING

In many cases, ML is used for solving problems without pre-defined models, which can be particularly useful when using ad hoc information for positioning. Inside positioning infrastructure, model-based methods are often more efficient than model-free methods. ML learns the model of the problem domain by itself based on the optimal information gain. In addition to positioning itself, ML can also be used for studying the problem and revealing the hidden dependencies between variables. For example, Wi-Fi fingerprint-based positioning is implemented with ML [223]. Nonlinear ensemble regression methods, such as random regression trees and deep learning, can be efficient in this domain [224], [225].

#### E. DEVELOPMENTS IN MISCELLANEOUS APPLICATIONS

With the rapid development of ML approaches in various sectors, there is a growing need to investigate and analyse ML suitability for localisation applications. In the case of location-based services, ML is growing in popularity as it can produce accurate positioning information for navigational purposes. An ML approach can support the achievement of better positioning performance both outdoors and indoors [226]. This positioning system allows real-time tracking and tracing of goods and enables the optimisation of logistics processes in many application areas [227]. In an indoor warehouse scenario, an ML algorithm allowed, for the first time, a monocular optical positioning application [227].

The authors of [228] proposed a visible light positioning (VLP) algorithm, which is based on ML and works to measure the relative distance between the receiving and transmitting ends of the camera. The combination of VLP systems and dual-function ML algorithms enables an increase in positioning accuracy by reducing the negative effect and eliminating low-intensity reflective signals [229]. In this approach, the position is determined by a proposed triangulation algorithm. The authors of [230] proposed an automated visual positioning system using deep learning, which aids to correctly place a workpiece on a fixture. This approach requires template matching across the image in which the template is compared with the local pixels.

ML methods are also used in fingerprint-based algorithms to enhance the precision and robustness of indoor positioning. The advancement of fingerprint localisation technology is a promising method for indoor positioning in various applications [231]. The authors of [232] developed a fingerprint-based localisation method, which is combined with ML and a heterogeneous feature fusion model. Another promising positioning method is smartphone-based indoor tracking, exploiting opportunistic sensing and machine learning techniques, e.g. SLAM. The SLAM method offers a new intelligent filtering approach to maintain good positioning performance [233].

#### F. SUMMARY

The research interest towards ML methods in positioning has been continuously increasing during the last five years [234]. ML facilitates the use of more data sources for positioning and development of collaborative positioning schemes, which are too complex for traditional methods. ML plays an important role in developing future location-based services by seamlessly using many available positioning infrastructure and other information sources [234].

#### **IX. DISCUSSION**

As UWB is a relatively new standard, the maturity in chip production is rapidly increasing. New accurate and affordable UWB hardware has been recently introduced by Decawave and NXP for industrial applications, and UWB chips are included in recent phone models and other consumer products by the biggest mobile phone manufacturers. This development has made UWB an attractive technology for indoor positioning purposes. Recent UWB positioning systems support decimetre-level accuracy and 60 m distance [105].

Most of the positioning algorithms used with UWB are the same, which have been used for positioning for decades, including multilateration, LS solution, Kalman filtering and particle filtering, but the increased popularity and lowering prices of UWB positioning are also attracting new application areas, solutions and algorithms. While traditional approaches are still optimal for the standard multilateration problem, different ML methods have been proposed to cover special situations, such as the use of heterogeneous data, detection of outliers, mitigation of NLOS situations and automatic adaptation to changing conditions. Automatically calibrated multirobot positioning systems with collaborative positioning and pose estimation methods have been proposed to assist in controlling industrial robots, drones or cars [159], [235]. Multisensor fusion has been studied for increasing the reliability and availability of the positioning service and ML methods to allow seamless roaming between positioning systems and the use of signals of opportunity together with actual positioning observables. Similarly, collaborative positioning can increase the positioning reliability under difficult conditions.

The recent literature in UWB positioning for smart logistics contains adaptation of UWB positioning to many new application areas and increasing the flexibility and reliability using new algorithms.

In the future, low-earth orbit (LEO) satellites can be used to provide an additional mode for both outdoor, and indoor navigation [236]. We intend to implement our proposed positioning system (mobile App), which involves the use of GNSSs, inertial sensors and UWB for smart logistics applications and researching the possibility of using LEO-based positioning observables fused with other available information.

#### **X. CONCLUSION**

Due to its versatility, accuracy and robustness, UWB technology is considered an efficient and reliable localisation method for implementing smart logistics. The wide adoption of UWB in location-based services confirms its ability to make an effective compromise among the cost, resource budget and precision. This paper summarises the most recent studies that have adopted UWB in sensitive applications that require high precision positioning in which UWB has successfully reduced the localisation error to a few centimetres. UWB advances for location provision are foreseen to grow as the adoption of this technology in mass-market devices increases via standards and related developments. UWB will likely not only be providing location solutions for special industrial applications but also consumer devices in the future. Moreover, this paper presents a compact review that focuses on using multi-sensor fusion-based systems in specific applications. We highlight the use of inertial sensors, remote sensing devices, visual sensors and other RF-based navigational sensors to achieve seamless navigation under various conditions.

Additionally, we present a higher perspective on the algorithms used for either single-sensor systems or multi-sensor fusion systems, such as Kalman filter, LS, PF, federated filter structure and ML methods. We have designed the architecture of this article to be a compact tutorial for researchers seeking an overall view on UWB positioning technology, aiming to make it a landmark article on the literature track of UWB.

#### **CONFLICT OF INTERESTS**

The authors declare no conflict of interests.

#### ACKNOWLEDGMENT

The authors would like to thank Prof. Miadreza Shafiekhah and the anonymous reviewers for the helpful remarks to further improve the manuscript.

#### REFERENCES

- S. Subedi and J.-Y. Pyun, "A survey of smartphone-based indoor positioning system using RF-based wireless technologies," *Sensors*, vol. 20, no. 24, p. 7230, Dec. 2020. [Online]. Available: https://www.mdpi.com/1424-8220/20/24/7230
- [2] D. B. Ninh, J. He, V. T. Trung, and D. P. Huy, "An effective random statistical method for indoor positioning system using WiFi fingerprinting," *Future Gener. Comput. Syst.*, vol. 109, pp. 238–248, Aug. 2020. [Online]. Available: https://www.sciencedirect.com/science/ article/pii/S0167739X19324835
- [3] C. E. A. Bundak, M. A. A. Rahman, M. K. A. Karim, and N. H. Osman, "Fuzzy rank cluster top k Euclidean distance and triangle based algorithm for magnetic field indoor positioning system," *Alexandria Eng. J.*, vol. 61, no. 5, pp. 3645–3655, May 2022. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1110016821005883
- [4] P. Bencak, D. Hercog, and T. Lerher, "Indoor positioning system based on Bluetooth low energy technology and a nature-inspired optimization algorithm," *Electronics*, vol. 11, no. 3, p. 308, Jan. 2022. [Online]. Available: https://www.mdpi.com/2079-9292/11/3/308
- [5] S. Jeong, J. Min, and Y. Park, "Indoor positioning using deep-learningbased pedestrian dead reckoning and optical camera communication," *IEEE Access*, vol. 9, pp. 133725–133734, 2021.
- [6] R. Vleugels, B. Van Herbruggen, J. Fontaine, and E. De Poorter, "Ultrawideband indoor positioning and IMU-based activity recognition for ice hockey analytics," *Sensors*, vol. 21, no. 14, p. 4650, Jul. 2021. [Online]. Available: https://www.mdpi.com/1424-8220/21/14/4650
- [7] Y. Song, F. R. Yu, L. Zhou, X. Yang, and Z. He, "Applications of the Internet of Things (IoT) in smart logistics: A comprehensive survey," *IEEE Internet Things J.*, vol. 8, no. 6, pp. 4250–4274, Mar. 2021, doi: 10.1109/jiot.2020.3034385.
- [8] W. Wu, L. Shen, Z. Zhao, M. Li, and G. Q. Huang, "Industrial IoT and long short-term memory network enabled genetic indoor tracking for factory logistics," *IEEE Trans. Ind. Informat.*, early access, Jan. 27, 2022, doi: 10.1109/TII.2022.3146598.
- [9] I. V. Pospelova, I. V. Cherepanova, D. S. Bragin, I. A. Sidorov, E. Y. Kostyuchenko, and V. N. Serebryakova, "The estimation of the potential for using smart-trackers as a part of a medical indoor-positioning system," *Electronics*, vol. 11, no. 1, p. 107, Dec. 2021. [Online]. Available: https://www.mdpi.com/2079-9292/11/1/107
- [10] A. Alarifi, A. Al-Salman, M. Alsaleh, A. Alnafessah, S. Al-Hadhrami, M. Al-Ammar, and H. Al-Khalifa, "Ultra wideband indoor positioning technologies: Analysis and recent advances," *Sensors*, vol. 16, no. 5, p. 707, May 2016, doi: 10.3390/s16050707.
- [11] J. Park, H. Kim, J. Yoon, H. Kim, C. Park, and D. Hong, "Development of an ultrasound technology-based indoor-location monitoring service system for worker safety in shipbuilding and offshore industry," *Processes*, vol. 9, no. 2, p. 304, Feb. 2021. [Online]. Available: https://www.mdpi.com/2227-9717/9/2/304
- [12] S. Woo, S. Jeong, E. Mok, L. Xia, C. Choi, M. Pyeon, and J. Heo, "Application of WiFi-based indoor positioning system for labor tracking at construction sites: A case study in Guangzhou MTR," *Automat. Construct.*, vol. 20, no. 1, pp. 3–13, Jan. 2011. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S092658051000107X

- [13] F. Nekoogar, Ultra-Wideband Communications: Fundamentals and Applications. Upper Saddle River, NJ, USA: Prentice-Hall, 2005.
- [14] P. Stephan, I. Heck, P. Krau, and G. Frey, "Evaluation of indoor positioning technologies under industrial application conditions in the Smart-FactoryKL based on EN ISO 9283," *IFAC Proc. Volumes*, vol. 42, no. 4, pp. 870–875, 2009. [Online]. Available: https://www.sciencedirect.com/ science/article/pii/S1474667016339039
- [15] S. Krishnan and R. X. M. Santos, "Real-time asset tracking for smart manufacturing," in *Implementing Industry 4.0* (Intelligent Systems Reference Library). Cham, Switzerland: Springer, 2021, pp. 25–53, doi: 10.1007/978-3-030-67270-6\_2.
- [16] L. Barbieri, M. Brambilla, A. Trabattoni, S. Mervic, and M. Nicoli, "UWB localization in a smart factory: Augmentation methods and experimental assessment," *IEEE Trans. Instrum. Meas.*, vol. 70, pp. 1–18, 2021.
- [17] K. Zhao, M. Zhu, B. Xiao, X. Yang, C. Gong, and J. Wu, "Joint RFID and UWB technologies in intelligent warehousing management system," *IEEE Internet Things J.*, vol. 7, no. 12, pp. 11640–11655, Dec. 2020, doi: 10.1109/jiot.2020.2998484.
- [18] J. A. Fernandez-Madrigal, E. Cruz-Martin, J. Gonzalez, C. Galindo, and J. L. Blanco, "Application of UWB and GPS technologies for vehicle localization in combined indoor-outdoor environments," in *Proc. 9th Int. Symp. Signal Process. Appl.*, Feb. 2007, pp. 1–4.
- [19] Y. Xianjia, L. Qingqing, J. P. Queralta, J. Heikkonen, and T. Westerlund, "Applications of UWB networks and positioning to autonomous robots and industrial systems," in *Proc. 10th Medit. Conf. Embedded Comput.* (*MECO*), 2021, pp. 1–6.
- [20] D. Minoli and B. Occhiogrosso, "Ultrawideband (UWB) technology for smart cities IoT applications," in *Proc. IEEE Int. Smart Cities Conf. (ISC)*, Sep. 2018, pp. 1–8.
- [21] R. S. Kulikov, "Integrated UWB/IMU system for high rate indoor navigation with cm-level accuracy," in *Proc. Moscow Workshop Electron. Netw. Technol. (MWENT)*, Mar. 2018, pp. 1–4, doi: 10.1109/mwent.2018.8337235.
- [22] J. Wang, M. Wang, D. Yang, F. Liu, and Z. Wen, "UWB positioning algorithm and accuracy evaluation for different indoor scenes," *Int. J. Image Data Fusion*, vol. 12, pp. 1–23, Jan. 2021, doi: 10.1080/19479832.2020.1864788.
- [23] Z.-G. Wu, C.-Y. Lin, H.-W. Chang, and P. T. Lin, "Inline inspection with an industrial robot (IIIR) for mass-customization production line," *Sensors*, vol. 20, no. 11, p. 3008, May 2020, doi: 10.3390/ s20113008.
- [24] P. T. Lin, C.-A. Liao, and S.-H. Liang, "Probabilistic indoor positioning and navigation (PIPN) of autonomous ground vehicle (AGV) based on wireless measurements," *IEEE Access*, vol. 9, pp. 25200–25207, 2021, doi: 10.1109/access.2021.3057415.
- [25] F. Mazhar, M. G. Khan, and B. Sällberg, "Precise indoor positioning using UWB: A review of methods, algorithms and implementations," *Wireless Pers. Commun.*, vol. 97, no. 3, pp. 4467–4491, Aug. 2017, doi: 10.1007/s11277-017-4734-x.
- [26] A. Yassin, Y. Nasser, M. Awad, A. Al-Dubai, R. Liu, C. Yuen, R. Raulefs, and E. Aboutanios, "Recent advances in indoor localization: A survey on theoretical approaches and applications," *IEEE Commun. Surveys Tuts.*, vol. 19, no. 2, pp. 1327–1346, 2nd Quart., 2017, doi: 10.1109/comst.2016.2632427.
- [27] P. Forum. Classification Criteria. Accessed: Jan. 26, 2021. [Online]. Available: https://julkaisufoorumi.fi/en/evaluations/classification-criteria
- [28] B. Feng and Q. Ye, "Operations management of smart logistics: A literature review and future research," *Frontiers Eng. Manage.*, vol. 8, no. 3, pp. 344–355, Apr. 2021, doi: 10.1007/s42524-021-0156-2.
- [29] L. Zwirello, T. Schipper, M. Harter, and T. Zwick, "UWB localization system for indoor applications: Concept, realization and analysis," *J. Electr. Comput. Eng.*, vol. 2012, pp. 1–11, May 2012, doi: 10.1155/2012/849638.
- [30] Y. Gu, A. Lo, and I. Niemegeers, "A survey of indoor positioning systems for wireless personal networks," *IEEE Commun. Surveys Tuts.*, vol. 11, no. 1, pp. 13–32, 1st Quart., 2009, doi: 10.1109/surv.2009. 090103.
- [31] P. Dabove, V. Di Pietra, M. Piras, A. A. Jabbar, and S. A. Kazim, "Indoor positioning using ultra-wide band (UWB) technologies: Positioning accuracies and sensors' performances," in *Proc. IEEE/ION Position, Location Navigat. Symp. (PLANS)*, Apr. 2018, pp. 175–184, doi: 10.1109/plans.2018.8373379.

- [32] Y.-C. Chen, A. I.-C. Lai, and R.-B. Wu, "UWB-assisted high-precision positioning in a UTM prototype," in *Proc. IEEE Topical Conf. Wireless Sensors Sensor Netw. (WiSNeT)*, Jan. 2020, pp. 42–45, doi: 10.1109/wisnet46826.2020.9037604.
- [33] A. Poulose, O. S. Eyobu, M. Kim, and D. S. Han, "Localization error analysis of indoor positioning system based on UWB measurements," in *Proc. 11th Int. Conf. Ubiquitous Future Netw. (ICUFN)*, Jul. 2019, pp. 84–88, doi: 10.1109/icufn.2019.8806041.
- [34] M. Li, H. Zhu, S. You, and C. Tang, "UWB-based localization system aided with inertial sensor for underground coal mine applications," *IEEE Sensors J.*, vol. 20, no. 12, pp. 6652–6669, Jun. 2020, doi: 10.1109/jsen.2020.2976097.
- [35] N. Zhang, "Smart logistics path for cyber-physical systems with Internet of Things," *IEEE Access*, vol. 6, pp. 70808–70819, 2018, doi: 10.1109/access.2018.2879966.
- [36] X. Tang, "Research on smart logistics model based on Internet of Things technology," *IEEE Access*, vol. 8, pp. 151150–151159, 2020, doi: 10.1109/access.2020.3016330.
- [37] M. Humayun, N. Jhanjhi, B. Hamid, and G. Ahmed, "Emerging smart logistics and transportation using IoT and blockchain," *IEEE Internet Things Mag.*, vol. 3, no. 2, pp. 58–62, Jun. 2020, doi: 10.1109/iotm.0001.1900097.
- [38] Y. Issaoui, A. Khiat, A. Bahnasse, and H. Ouajji, "Smart logistics: Study of the application of blockchain technology," *Proc. Comput. Sci.*, vol. 160, pp. 266–271, Jan. 2019, doi: 10.1016/j.procs.2019.09.467.
- [39] Y. Su and Q.-M. Fan, "The green vehicle routing problem from a smart logistics perspective," *IEEE Access*, vol. 8, pp. 839–846, 2020, doi: 10.1109/access.2019.2961701.
- [40] Y. Issaoui, A. Khiat, A. Bahnasse, and H. Ouajji, "Toward smart logistics: Engineering insights and emerging trends," *Arch. Comput. Methods Eng.*, vol. 28, no. 4, pp. 3183–3210, Sep. 2020, doi: 10.1007/s11831-020-09494-2.
- [41] N. Nantee and P. Sureeyatanapas, "The impact of logistics 4.0 on corporate sustainability: A performance assessment of automated warehouse operations," *Benchmarking, Int. J.*, vol. 28, no. 10, pp. 2865–2895, Mar. 2021, doi: 10.1108/bij-11-2020-0583.
- [42] S. Lu, C. Xu, R. Y. Zhong, and L. Wang, "A RFID-enabled positioning system in automated guided vehicle for smart factories," *J. Manuf. Syst.*, vol. 44, pp. 179–190, Jul. 2017, doi: 10.1016/j.jmsy.2017.03.009.
- [43] Y. Tao and L. Zhao, "A novel system for WiFi radio map automatic adaptation and indoor positioning," *IEEE Trans. Veh. Technol.*, vol. 67, no. 11, pp. 10683–10692, Nov. 2018, doi: 10.1109/tvt.2018.2867065.
- [44] S. Horsmanheimo, S. Lembo, L. Tuomimaki, S. Huilla, P. Honkamaa, M. Laukkanen, and P. Kemppi, "Indoor positioning platform to support 5G location based services," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, May 2019, pp. 1–6, doi: 10.1109/ICCW.2019.8757118.
- [45] A. Rácz-Szabó, T. Ruppert, L. Bántay, A. Löcklin, L. Jakab, and J. Abonyi, "Real-time locating system in production management," *Sensors*, vol. 20, no. 23, p. 6766, Nov. 2020, doi: 10.3390/s20236766.
- [46] Z. Silvia, C. Martina, S. Fabio, and P. Alessandro, "Ultra wide band indoor positioning system: Analysis and testing of an IPS technology," *IFAC-PapersOnLine*, vol. 51, no. 11, pp. 1488–1492, 2018, doi: 10.1016/j.ifacol.2018.08.292.
- [47] A. Motroni, A. Buffi, and P. Nepa, "A survey on indoor vehicle localization through RFID technology," *IEEE Access*, vol. 9, pp. 17921–17942, 2021, doi: 10.1109/access.2021.3052316.
- [48] F. Bernardini, A. Buffi, D. Fontanelli, D. Macii, V. Magnago, M. Marracci, A. Motroni, P. Nepa, and B. Tellini, "Robot-based indoor positioning of UHF-RFID tags: The SAR method with multiple trajectories," *IEEE Trans. Instrum. Meas.*, vol. 70, pp. 1–15, 2021, doi: 10.1109/tim.2020.3033728.
- [49] R. Montoliu, E. Sansano, A. Gascó, O. Belmonte, and A. Caballer, "Indoor positioning for monitoring older adults at home: Wi-Fi and BLE technologies in real scenarios," *Electronics*, vol. 9, no. 5, p. 728, Apr. 2020, doi: 10.3390/electronics9050728.
- [50] Z. Zou, "Impulse radio UWB for the Internet-of-Things: A study on UHF/UWB hybrid solution," Ph.D. dissertation, KTH School Inf. Commun. Technol., Stockholm, Sweden, 2011.
- [51] M. Ridolfi, S. Vandermeeren, J. Defraye, H. Steendam, J. Gerlo, D. D. Clercq, J. Hoebeke, and E. D. Poorter, "Experimental evaluation of UWB indoor positioning for sport postures," *Sensors*, vol. 18, no. 1, p. 168, Jan. 2018, doi: 10.3390/s18010168.

- [52] P. Helo and A. H. M. Shamsuzzoha, "Real-time supply chain— A blockchain architecture for project deliveries," *Robot. Comput.-Integr. Manuf.*, vol. 63, Jun. 2020, Art. no. 101909, doi: 10.1016/j.rcim.2019.101909.
- [53] C.-H. Cheng, Y.-H. Kuo, H. Lam, and M. Petering, "Real-time locationpositioning technologies for managing cart operations at a distribution facility," *Appl. Sci.*, vol. 11, no. 9, p. 4049, Apr. 2021, doi: 10.3390/app11094049.
- [54] L. Barbieri, M. Brambilla, A. Trabattoni, S. Mervic, and M. Nicoli, "UWB localization in a smart factory: Augmentation methods and experimental assessment," *IEEE Trans. Instrum. Meas.*, vol. 70, pp. 1–18, 2021, doi: 10.1109/tim.2021.3074403.
- [55] R. Sonderleittner. (2020). Why Real-Time Tracking is Part of the Smart Factory. [Online]. Available: https://www.eliko.ee/why-real-timetracking-is-part-of-the-smart-factory/
- [56] N. Zhu, J. Marais, D. Bétaille, and M. Berbineau, "GNSS position integrity in urban environments: A review of literature," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 9, pp. 2762–2778, Sep. 2018.
- [57] K. K. Muthukrishnan, N. Meratnia, and M. Lijding, "FLAVOUR: Friendly location-aware conference assistant with privacy observant architecture," Centre Telematics Inf. Technol. (CTIT), Enschede, The Netherlands, CTIT Tech. Rep., Jun. 2005, nos. 5–28. [Online]. Available: https://ris.utwente.nl/ws/portalfiles/portal/5398270/icsocDemo.pdf
- [58] Z. Xu, S. Huang, and J. Ding, "A new positioning method for indoor laser navigation on under-determined condition," in *Proc. 6th Int. Conf. Instrum. Meas., Comput., Commun. Control (IMCCC)*, Jul. 2016, pp. 703–706, doi: 10.1109/imccc.2016.158.
- [59] Q. Zou, Q. Sun, L. Chen, B. Nie, and Q. Li, "A comparative analysis of LiDAR SLAM-based indoor navigation for autonomous vehicles," *IEEE Trans. Intell. Transp. Syst.*, early access, Mar. 18, 2021, doi: 10.1109/tits.2021.3063477.
- [60] G. Li, X. Liao, H. Huang, S. Song, B. Liu, and Y. Zeng, "Robust stereo visual SLAM for dynamic environments with moving object," *IEEE Access*, vol. 9, pp. 32310–32320, 2021, doi: 10.1109/access.2021.3059866.
- [61] V. Honkavirta, T. Perala, S. Ali-Loytty, and R. Piche, "A comparative survey of WLAN location fingerprinting methods," in *Proc. 6th Workshop Positioning, Navigat. Commun.*, Mar. 2009, pp. 243–251, doi: 10.1109/wpnc.2009.4907834.
- [62] R. Faragher and R. Harle, "Location fingerprinting with Bluetooth low energy beacons," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 11, pp. 2418–2428, Nov. 2015, doi: 10.1109/jsac.2015.2430281.
- [63] F. Zafari, A. Gkelias, and K. K. Leung, "A survey of indoor localization systems and technologies," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 3, pp. 2568–2599, 3rd Quart., 2019.
- [64] J. Gjengset, J. Xiong, G. McPhillips, and K. Jamieson, "Phaser: Enabling phased array signal processing on commodity WiFi access points," in *Proc. 20th Annu. Int. Conf. Mobile Comput. Netw. (MobiCom).* New York, NY, USA: Association for Computing Machinery, Sep. 2014, pp. 153–164, doi: 10.1145/2639108.2639139.
- [65] W. Komamiya, S. Tang, and S. Obana, "Precise angle estimation by jointly using spatial/temporal change of channel state information and its application in pedestrian positioning," *IEEE Access*, vol. 9, pp. 59420–59431, 2021.
- [66] C. Gentner, M. Ulmschneider, I. Kuehner, and A. Dammann, "WiFi-RTT indoor positioning," in *Proc. IEEE/ION Position, Location Navigat. Symp. (PLANS)*, Apr. 2020, pp. 1029–1035, doi: 10.1109/plans46316.2020.9110232.
- [67] L. Botler, M. Spork, K. Diwold, and K. Romer, "Direction finding with UWB and BLE: A comparative study," in *Proc. IEEE 17th Int. Conf. Mobile Ad Hoc Sensor Syst. (MASS)*, Dec. 2020, doi: pp. 44–52, 10.1109/mass50613.2020.00016.
- [68] J. Talvitie, M. Koivisto, T. Levanen, M. Valkama, G. Destino, and H. Wymeersch, "High-accuracy joint position and orientation estimation in sparse 5G mmWave channel," in *Proc. IEEE Int. Conf. Commun.* (*ICC*), May 2019, pp. 1–7, doi: 10.1109/icc.2019.8761910.
- [69] A. Wilzeck, M. P. Guirao, and E. Dimitrov, "White paper on UWB technology and regulation," wiseSense GmbH, Hannover, Germany, Tech. Rep., 2018.
- [70] F. Sabath, E. L. Mokole, and S. N. Samaddar, "Definition and classification of ultra-wideband signals and devices," *URSI Radio Sci. Bull.*, vol. 2005, no. 313, pp. 12–26, 2005.
- [71] A. Muqaibel, A. Safaai-Jazi, and S. Riad, "Ultra wideband vs. narrowband communications," in *Proc. Riyadh Int. Commun. IT Conf.*, Feb. 2004.

- [72] Federal Communications Commission. (2002). Revision of Part 15 of the Commission's Rules Regarding Ultra-Wideband Transmission Systems. [Online]. Available: https://transition.fcc.gov/ Bureaus/Engineering\_Technology/Orders/2002/fcc02048.pdf
- [73] M. Matin, "Ultra wideband preliminaries," in *Ultra Wideband*. London, U.K.: IntechOpen, Aug. 2010, doi: 10.5772/10059.
- [74] J.-S. Lee, Y.-W. Su, and C.-C. Shen, "A comparative study of wireless protocols: Bluetooth, UWB, ZigBee, and Wi-Fi," in *Proc. 33rd Annu. Conf. IEEE Ind. Electron. Soc. (IECON)*, Nov. 2007, pp. 46–51, doi: 10.1109/iecon.2007.4460126.
- [75] Z. Yin, X. Jiang, Z. Yang, N. Zhao, and Y. Chen, "WUB-IP: A highprecision UWB positioning scheme for indoor multiuser applications," *IEEE Syst. J.*, vol. 13, no. 1, pp. 279–288, Mar. 2019, doi: 10.1109/jsyst.2017.2766690.
- [76] D. Dardari, A. Conti, U. Ferner, A. Giorgetti, and M. Z. Win, "Ranging with ultrawide bandwidth signals in multipath environments," *Proc. IEEE*, vol. 97, no. 2, pp. 404–426, Feb. 2009, doi: 10.1109/JPROC.2008.2008846.
- [77] L. E. Miller, "Why UWB? A review of ultrawideband technology, report to NETEX project office, DARPA," Nat. Inst. Standards Technol. (NIST), U.S. Dept. Commerce, Washington, DC, USA, Tech. Rep., 2003. [Online]. Available: https://citeseerx.ist.psu.edu/ viewdoc/download?doi=10.1.1.76.7051&rep=rep1&type=pdf
- [78] M. Santhanam, "UWB technology and its applications—A survey," M.S. thesis, School Eng., Jönköping Univ., Jönköping, Sweden, 2011.
- [79] M. Ridolfi, S. Van de Velde, H. Steendam, and E. De Poorter, "Analysis of the scalability of UWB indoor localization solutions for high user densities," *Sensors*, vol. 18, no. 6, p. 1875, Jun. 2018, doi: 10.3390/s18061875.
- [80] K. Khurshid and I. A. Khokhar, "Comparison survey of 4G competitors (OFDMA, MC CDMA, UWB, IDMA)," in *Proc. Int. Conf. Aerosp. Sci. Eng. (ICASE)*, Aug. 2013, pp. 1–7, doi: 10.1109/icase.2013.6785555.
- [81] G. Irene and A. Rajesh, "A penta-band reject inside cut koch fractal hexagonal monopole UWB MIMO antenna for portable devices," *Prog. Electromagn. Res.*, vol. 82, pp. 225–235, 2018. [Online]. Available: https://www.jpier.org/PIERC/view/18020604/
- [82] FCC. (2002). Rules and Regulations of Ultra-Wideband Operation. [Online]. Available: https://www.govinfo.gov/content/pkg/FR-2002-05-16/pdf/02-11929.pdf#page=5
- [83] European Telecommunications Standards Institute. (2016). Short Range Devices (SRD) Using Ultra Wide Band Technology (UWB); Harmonised Standard Covering the Essential Requirements of Article 3.2 of the Directive 2014/53/EU; Part 1: Requirements for Generic UWB Applications. [Online]. Available: https://www.etsi. org/deliver/etsi\_en/302000\_302099/30206501/02.01.01\_60/en\_ 30206501v020101p.pdf
- [84] (2019). Regulatory Requirements for Ultra-Wideband Technology. [Online]. Available: https://www.cetecom.com/en/news/regulatoryrequirements-for-ultra-wideb and-technology/
- [85] European Telecommunications Standards Institute. (2016). Short Range Devices (SRD) Using Ultra Wide Band Technology (UWB); Harmonised Standard Covering the Essential Requirements of Article 3.2 of the Directive 2014/53/EU; Part 2: Requirements for UWB Location Tracking. [Online]. Available: https://www.etsi.org/deliver/ etsi\_en/302000\_302099/30206502/02.01.01\_60/en\_30206502v020101p. pdf
- [86] About FiRa Consortium. Accessed: Jun. 1, 2021. [Online]. Available: https://www.firaconsortium.org/about/consortium
- [87] IEEE Standard for Low-Rate Wireless Networks-Amendment 1: Enhanced Ultra Wideband (UWB) Physical Layers (PHYs) and Associated Ranging Techniques, IEEE Standard 802.15.4z-2020 (Amendment to IEEE Std 802.15.4-2020), 2020, pp. 1–174. [Online]. Available: https://ieeexplore.ieee.org/document/9179124
- [88] Qorvo. (2020). Qorvo Completes Acquisition of Decawave. [Online]. Available: https://www.qorvo.com/newsroom/news/2020/qorvocompletes-acquisition-of-decawave
- [89] A. R. J. Ruiz and F. S. Granja, "Comparing ubisense, bespoon, and decawave UWB location systems: Indoor performance analysis," *IEEE Trans. Instrum. Meas.*, vol. 66, no. 8, pp. 2106–2117, Aug. 2017.
- [90] J. Kulmer, S. Hinteregger, B. Grosswindhager, M. Rath, M. S. Bakr, E. Leitinger, and K. Witrisal, "Using DecaWave UWB transceivers for high-accuracy multipath-assisted indoor positioning," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, May 2017, pp. 1239–1245. [Online]. Available: https://ieeexplore.ieee.org/document/7962828

- [91] Q. Tian, K. I.-K. Wang, and Z. Salcic, "An INS and UWB fusion approach with adaptive ranging error mitigation for pedestrian tracking," *IEEE Sensors J.*, vol. 20, no. 8, pp. 4372–4381, Apr. 2020.
- [92] I. Dotlic, A. Connell, H. Ma, J. Clancy, and M. McLaughlin, "Angle of arrival estimation using decawave DW1000 integrated circuits," in *Proc. 14th Workshop Positioning, Navigat. Commun. (WPNC)*, Oct. 2017, pp. 1–6, doi: 10.1109/wpnc.2017.8250079.
- [93] M. Cimdins, S. O. Schmidt, and H. Hellbrück, "MAMPI-UWB— Multipath-assisted device-free localization with magnitude and phase information with UWB transceivers," *Sensors*, vol. 20, no. 24, p. 7090, Dec. 2020.
- [94] Pozyx. (2021). Positioning Mode. [Online]. Available: https://docs.pozyx.io/enterprise/Positioning.1224016064.html
- [95] (2021). Exafore. [Online]. Available: https://exafore.com
- [96] Sewio. (2021). Time Difference of Arrival. [Online]. Available: https://www.sewio.net/uwb-technology/time-difference-of-arrival/
- [97] Eliko. (2021). Next-Generation Location Tracking. [Online]. Available: https://www.eliko.ee/
- [98] NXP. (2022). Secure Ultra-Wideband (UWB). [Online]. Available: https://www.nxp.com/products/wireless/secure-ultra-wideband-uwb: UWB-TRIMENSION
- [99] Apple. (2019). Nearby Interactions With U1. [Online]. Available: https://www.apple.com/fi/newsroom/2019/09/iphone-11-pro-andiphone-11-pro-max-the-most-powerful-and-advanced-smartphones
- [100] Samsung. (2019). Key Industry Players the ASSA ABLOY Group, HID, NXP, Samsung, Bosch, Sony, LitePoint and TTA Establish FiRa Consortium to Drive Seamless User Experiences Using Ultra-Wideband Technology. [Online]. Available: https://news.samsung.com/global/keyindustry-players-the-assa-abloy-group-hid-nxp-samsung-bosch-sonylitepoint-and-tta-establish-fira-consortium-to-drive-seamless-userexperiences-using-ultra-wideband-technology
- [101] Apple. (2021). iPhone 11 Pro and iPhone 11 Pro Max: The Most Powerful and Advanced Smartphones. [Online]. Available: https://developer.apple.com/nearby-interaction/
- [102] Bosch. (2021). Perfectly Keyless. [Online]. Available: https://www.boschmobility-solutions.com/en/solutions/software-and-services/perfectlykeyless/
- [103] P. Carbone, A. Cazzorla, P. Ferrari, A. Flammini, A. Moschitta, S. Rinaldi, T. Sauter, and E. Sisinni, "Low complexity UWB radios for precise wireless sensor network synchronization," *IEEE Trans. Instrum. Meas.*, vol. 62, no. 9, pp. 2538–2548, Sep. 2013, doi: 10.1109/tim.2013.2259101.
- [104] H. Soganci, S. Gezici, and H. Poor, "Accurate positioning in ultrawideband systems," *IEEE Wireless Commun.*, vol. 18, no. 2, pp. 19–27, Apr. 2011, doi: 10.1109/mwc.2011.5751292.
- [105] W. Shule, C. M. Almansa, J. P. Queralta, Z. Zou, and T. Westerlund, "UWB-based localization for multi-UAV systems and collaborative heterogeneous multi-robot systems," *Proc. Comput. Sci.*, vol. 175, pp. 357–364, Jan. 2020, doi: 10.1016/j.procs.2020.07.051.
- [106] S. P. Rana, M. Dey, H. U. Siddiqui, G. Tiberi, M. Ghavami, and S. Dudley, "UWB localization employing supervised learning method," in *Proc. IEEE 17th Int. Conf. Ubiquitous Wireless Broadband (ICUWB)*, Sep. 2017, pp. 1–5, doi: 10.1109/icuwb.2017.8250971.
- [107] H. Zhang, X. Liu, T. A. Gulliver, and X.-R. Cui, "AOA estimation for UWB positioning using a mono-station antenna array," *J. Electron. Inf. Technol.*, vol. 35, no. 8, pp. 2024–2028, Feb. 2014, doi: 10.3724/sp.j.1146.2012.01639.
- [108] X. D. Tu, H. Zhang, X. R. Cui, and X. Liu, "An improved mono-station UWB TOA/AOA positioning method," *Appl. Mech. Mater.*, vols. 229–231, pp. 1373–1376, Nov. 2012, doi: 10.4028/www.scientific.net/amm.229-231.1373.
- [109] M. Stone. (Jan. 2021). What is Ultra-Wideband Technology and How Does it Work? [Online]. Available: https://insights. samsung.com/2021/01/25/what-is-ultra-wideband-and-how-does-itwork-2/
- [110] M. Martalo, S. Perri, G. Verdano, F. De Mola, F. Monica, and G. Ferrari, "Improved UWB TDoA-based positioning using a single hotspot for industrial IoT applications," *IEEE Trans. Ind. Informat.*, vol. 18, no. 6, pp. 3915–3925, Jun. 2022.
- [111] D. Neirynck, E. Luk, and M. McLaughlin, "An alternative double-sided two-way ranging method," in *Proc. 13th Workshop Positioning, Navigat. Commun. (WPNC)*, Oct. 2016, pp. 1–4, doi: 10.1109/wpnc.2016.7822844.

- [112] Y. Cheng and T. Zhou, "UWB indoor positioning algorithm based on TDOA technology," in *Proc. 10th Int. Conf. Inf. Technol. Med. Educ.* (*ITME*), Aug. 2019, pp. 777–782, doi: 10.1109/itme.2019.00177.
- [113] R. Zandian and U. Witkowski, "Implementation challenges of synchronisation of UWB nodes in TDoA structures," in *Proc. Int. Conf. Indoor Positioning Indoor Navigat. (IPIN)*, Sep. 2018, pp. 1–8.
- [114] J. Kolakowski, J. Cichocki, P. Makal, and R. Michnowski, "An ultrawideband system for vehicle positioning," *Int. J. Electron. Telecommun.*, vol. 56, no. 3, pp. 247–256, Sep. 2010, doi: 10.2478/v10177-010-0032-1.
- [115] P. Corbalán, G. P. Picco, and S. Palipana, "Chorus: UWB concurrent transmissions for GPS-like passive localization of countless targets," in *Proc. 18th Int. Conf. Inf. Process. Sensor Netw.*, Apr. 2019, pp. 133–144.
- [116] S. Leugner, M. Pelka, and H. Hellbruck, "Comparison of wired and wireless synchronization with clock drift compensation suited for U-TDoA localization," in *Proc. 13th Workshop Positioning, Navigat. Commun.* (WPNC), Oct. 2016, pp. 1–4, doi: 10.1109/wpnc.2016.7822846.
- [117] S. Pittet, V. Renaudin, B. Merminod, and M. Kasser, "UWB and MEMS based indoor navigation," *J. Navigat.*, vol. 61, no. 3, pp. 369–384, Jun. 2008, doi: 10.1017/s0373463308004797.
- [118] J. Blazek, J. Jiranek, and J. Bajer, "Indoor passive positioning technique using ultra wide band modules," in *Proc. Int. Conf. Mil. Technol. (ICMT)*, May 2019, pp. 1–5.
- [119] M. Youssef, M. Mah, and A. Agrawala, "Challenges: Device-free passive localization for wireless environments," in *Proc. 13th Annu. ACM Int. Conf. Mobile Comput. Netw. (MobiCom).* New York, NY, USA: Association for Computing Machinery, Sep. 2007, pp. 222–229.
- [120] T. J. Daim and R. M. A. Lee, "A weighted least squares consideration for IR-UWB radar based device-free object positioning estimation for indoor environment," *Indonesian J. Electr. Eng. Comput. Sci.*, vol. 15, no. 2, p. 894, Aug. 2019, doi: 10.11591/ijeecs.v15.i2.pp894-901.
- [121] C.-S. Chen, "A non-line-of-sight error mitigation method for location estimation," *Int. J. Distrib. Sensor Netw.*, vol. 13, no. 1, Jan. 2017, Art. no. 155014771668273, doi: 10.1177/1550147716682739.
- [122] J. J. Caffery, "A new approach to the geometry of TOA location," in Proc. Veh. Technol. Conf. Fall., IEEE VTS Fall VTC. 52nd Veh. Technol. Conf., 2000, pp. 1943–1949, doi: 10.1109/vetecf.2000.886153.
- [123] Y. Zhang, J. Ren, and W. Chen, "A ToA-based location algorithm reducing the NLoS error under location-aware networks," in *Proc. 7th Int. Conf. Wireless Commun., Netw. Mobile Comput.*, Sep. 2011, pp. 1–4, doi: 10.1109/wicom.2011.6040577.
- [124] Y. T. Chan and K. C. Ho, "A simple and efficient estimator for hyperbolic location," *IEEE Trans. Signal Process.*, vol. 42, no. 8, pp. 1905–1915, Aug. 1994, doi: 10.1109/78.301830.
- [125] X. Wang, Z. Huang, F. Zheng, and X. Tian, "The research of indoor threedimensional positioning algorithm based on ultra-wideband technology," in *Proc. 39th Chin. Control Conf. (CCC)*, Jul. 2020, pp. 5144–5149, doi: 10.23919/ccc50068.2020.9189061.
- [126] V. Fox, J. Hightower, L. Liao, D. Schulz, and G. Borriello, "Bayesian filtering for location estimation," *IEEE Pervasive Comput.*, vol. 2, no. 3, pp. 24–33, Jul. 2003.
- [127] R. E. Kalman, "A new approach to linear filtering and prediction problems," J. Fluids Eng., vol. 82, no. 1, pp. 35–45, Mar. 1960, doi: 10.1115/1.3662552.
- [128] G. A. Darbellay, "An estimator of the mutual information based on a criterion for conditional independence," *Comput. Statist. Data Anal.*, vol. 32, no. 1, pp. 1–17, Nov. 1999, doi: 10.1016/s0167-9473%2899%2900020-1.
- [129] G. Schalk and E. C. Leuthardt, "Brain-computer interfaces using electrocorticographic signals," *IEEE Rev. Biomed. Eng.*, vol. 4, pp. 140–154, 2011, doi: 10.1109/rbme.2011.2172408.
- [130] R. Faragher, "Understanding the basis of the Kalman filter via a simple and intuitive derivation [lecture notes]," *IEEE Signal Process. Mag.*, vol. 29, no. 5, pp. 128–132, Sep. 2012, doi: 10.1109/msp.2012.2203621.
- [131] J. Hartikainen, A. Solin, and S. Särkkä, Optimal Filtering With Kalman Filters and Smoothers—A Manual for MATLAB Toolbox EKF/UKF, Aalto Univ., Espoo, Finland, 2011. [Online]. Available: http://citeseerx.ist.psu.edu/viewdoc/summary?doi=10.1.1.331.150
- [132] Y. Bar-Shalom, X. R. Li, and T. Kirubarajan, *Estimation With Applica*tions to Tracking and Navigation. Hoboken, NJ, USA: Wiley, 2001.
- [133] Y. Bar-Shalom, "Recursive tracking algorithms: From the Kalman filter to intelligent trackers for cluttered environment," in *Proc. IEEE Int. Conf. Control Appl. (ICCON)*, Apr. 1989, pp. 675–680, doi: 10.1109/iccon.1989.770605.
- [134] S. J. Julier and J. K. Uhlmann, "Unscented filtering and nonlinear estimation," *Proc. IEEE*, vol. 92, no. 3, pp. 401–422, Mar. 2004, doi: 10.1109/JPROC.2003.823141.

- [135] S. Sarkka, "On unscented Kalman filtering for state estimation of continuous-time nonlinear systems," *IEEE Trans. Autom. Control*, vol. 52, no. 9, pp. 1631–1641, Sep. 2007, doi: 10.1109/TAC.2007.904453.
- [136] I. Ullah, Y. Shen, X. Su, C. Esposito, and C. Choi, "A localization based on unscented Kalman filter and particle filter localization algorithms," *IEEE Access*, vol. 8, pp. 2233–2246, 2020, doi: 10.1109/access.2019.2961740.
- [137] D. Feng, C. Wang, C. He, Y. Zhuang, and X.-G. Xia, "Kalman-filterbased integration of IMU and UWB for high-accuracy indoor positioning and navigation," *IEEE Internet Things J.*, vol. 7, no. 4, pp. 3133–3146, Apr. 2020, doi: 10.1109/jiot.2020.2965115.
- [138] W. You, F. Li, L. Liao, and M. Huang, "Data fusion of UWB and IMU based on unscented Kalman filter for indoor localization of quadrotor UAV," *IEEE Access*, vol. 8, pp. 64971–64981, 2020, doi: 10.1109/access.2020.2985053.
- [139] P. Li, Y. Xu, T. Shen, and S. Bi, "INS/UWB integrated AGV localization employing Kalman filter for indoor LOS/NLOS mixed environment," in *Proc. Int. Conf. Adv. Mech. Syst. (ICAMechS)*, Aug. 2019, pp. 294–298, doi: 10.1109/icamechs.2019.8861620.
- [140] L. Cheng, Z. Wu, B. Lai, Q. Yang, A. Zhao, and Y. Wang, "Ultra wideband indoor positioning system based on artificial intelligence techniques," in *Proc. IEEE 21st Int. Conf. Inf. Reuse Integr. Data Sci. (IRI)*, Aug. 2020, pp. 438–444, doi: 10.1109/iri49571.2020.00073.
- [141] L. Cheng, H. Chang, K. Wang, and Z. Wu, "Real time indoor positioning system for smart grid based on UWB and artificial intelligence techniques," in *Proc. IEEE Conf. Technol. Sustainability (SusTech)*, Apr. 2020, pp. 1–7, doi: 10.1109/sustech47890.2020.9150486.
- [142] C. He, C. Tang, and C. Yu, "A federated derivative cubature Kalman filter for IMU-UWB indoor positioning," *Sensors*, vol. 20, no. 12, p. 3514, Jun. 2020, doi: 10.3390/s20123514.
- [143] G. De Angelis, A. Moschitta, and P. Carbone, "Positioning techniques in indoor environments based on stochastic modeling of UWB roundtrip-time measurements," *IEEE Trans. Intell. Transp. Syst.*, vol. 17, no. 8, pp. 2272–2281, Aug. 2016, doi: 10.1109/tits.2016.2516822.
- [144] N. Dwek, M. Birem, K. Geebelen, E. Hostens, A. Mishra, J. Steckel, and R. Yudanto, "Improving the accuracy and robustness of ultrawideband localization through sensor fusion and outlier detection," *IEEE Robot. Autom. Lett.*, vol. 5, no. 1, pp. 32–39, Jan. 2020, doi: 10.1109/Ira.2019.2943821.
- [145] T. Otim, L. E. Díez, A. Bahillo, P. Lopez-Iturri, and F. Falcone, "Effects of the body wearable sensor position on the UWB localization accuracy," *Electronics*, vol. 8, no. 11, p. 1351, Nov. 2019, doi: 10.3390/electronics8111351.
- [146] Y. Xu, Y. S. Shmaliy, Y. Li, and X. Chen, "UWB-based indoor human localization with time-delayed data using EFIR filtering," *IEEE Access*, vol. 5, pp. 16676–16683, 2017, doi: 10.1109/access.2017.2743213.
- [147] B. Zhu, X. Tao, J. Zhao, M. Ke, H. Wang, and W. Deng, "An integrated GNSS/UWB/DR/VMM positioning strategy for intelligent vehicles," *IEEE Trans. Veh. Technol.*, vol. 69, no. 10, pp. 10842–10853, Oct. 2020, doi: 10.1109/tvt.2020.3014516.
- [148] Y. Zhang, X. Tan, and C. Zhao, "UWB/INS integrated pedestrian positioning for robust indoor environments," *IEEE Sensors J.*, vol. 20, no. 23, pp. 14401–14409, Dec. 2020, doi: 10.1109/jsen.2020.2998815.
- [149] M. Strohmeier, T. Walter, J. Rothe, and S. Montenegro, "Ultra-wideband based pose estimation for small unmanned aerial vehicles," *IEEE Access*, vol. 6, pp. 57526–57535, 2018, doi: 10.1109/access.2018.2873571.
- [150] F. Wu and Z. Liu, "Research on UWB/IMU fusion positioning technology in mine," in *Proc. Int. Conf. Intell. Transp., Big Data Smart City* (*ICITBS*), Jan. 2020, pp. 934–937, doi: 10.1109/icitbs49701.2020.00207.
- [151] H. J. Kim, Y. Xie, H. Yang, C. Lee, and T. L. Song, "An efficient indoor target tracking algorithm using TDOA measurements with applications to ultra-wideband systems," *IEEE Access*, vol. 7, pp. 91435–91445, 2019.
- [152] M. Vemula, M. F. Bugallo, and P. M. Djuric, "Performance comparison of Gaussian-based filters using information measures," *IEEE Signal Process. Lett.*, vol. 14, no. 12, pp. 1020–1023, Dec. 2007.
- [153] G. Kitagawa, "A self-organizing state-space model," J. Amer. Stat. Assoc., vol. 93, no. 443, p. 1203, Sep. 1998, doi: 10.2307/2669862.
- [154] A. Haggenmiller, M. Krogius, and E. Olson, "Non-parametric error modeling for ultra-wideband localization networks," in *Proc. Int. Conf. Robot. Automat. (ICRA)*, May 2019, pp. 2568–2574.
- [155] Q. Tian, K. I.-K. Wang, and Z. Salcic, "A resetting approach for INS and UWB sensor fusion using particle filter for pedestrian tracking," *IEEE Trans. Instrum. Meas.*, vol. 69, no. 8, pp. 5914–5921, Aug. 2020.

- [156] Q. Tian, K. I.-K. Wang, and Z. Salcic, "A low-cost INS and UWB fusion pedestrian tracking system," *IEEE Sensors J.*, vol. 19, no. 10, pp. 3733–3740, May 15, 2019.
- [157] S. Zhao, B. Huang, and F. Liu, "Localization of indoor mobile robot using minimum variance unbiased FIR filter," *IEEE Trans. Autom. Sci. Eng.*, vol. 15, no. 2, pp. 410–419, Apr. 2018.
- [158] X. Fu and Y. Jia, "An improvement on resampling algorithm of particle filters," *IEEE Trans. Signal Process.*, vol. 58, no. 10, pp. 5414–5420, Oct. 2010, doi: 10.1109/tsp.2010.2053031.
- [159] S. Thrun, D. Fox, W. Burgard, and F. Dellaert, "Robust Monte Carlo localization for mobile robots," *Artif. Intell.*, vol. 128, nos. 1–2, pp. 99–141, May 2001, doi: 10.1016/s0004-3702%2801%2900069-8.
- [160] C. Xu, X. Wang, S. Duan, and J. Wan, "Spatial-temporal constrained particle filter for cooperative target tracking," *J. Netw. Comput. Appl.*, vol. 176, Feb. 2021, Art. no. 102913. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1084804520303726
- [161] S. Lenser and M. Veloso, "Sensor resetting localization for poorly modelled mobile robots," in *Proc. Millennium Conf. IEEE Int. Conf. Robot. Automat. Symp. (ICRA)*, Apr. 2000, pp. 1225–1232, doi: 10.1109/robot.2000.844766.
- [162] X. Zhu, J. Yi, J. Cheng, and L. He, "Adapted error map based mobile robot UWB indoor positioning," *IEEE Trans. Instrum. Meas.*, vol. 69, no. 9, pp. 6336–6350, Sep. 2020.
- [163] Q. Tian, K. I.-K. Wang, and Z. Salcic, "Human body shadowing effect on UWB-based ranging system for pedestrian tracking," *IEEE Trans. Instrum. Meas.*, vol. 68, no. 10, pp. 4028–4037, Oct. 2019.
- [164] S. Vandermeeren and H. Steendam, "PDR/UWB based positioning of a shopping cart," *IEEE Sensors J.*, vol. 21, no. 9, pp. 10864–10878, May 2021.
- [165] N. Sünderhauf and P. Protzel, "Towards robust graphical models for GNSS-based localization in urban environments," in *Proc. Int. Multi-Conf. Syst., Signals Devices*, Mar. 2012, pp. 1–6.
- [166] V. Indelman, S. Williams, M. Kaess, and F. Dellaert, "Factor graph based incremental smoothing in inertial navigation systems," in *Proc. 15th Int. Conf. Inf. Fusion*, Jul. 2012, pp. 2154–2161.
- [167] T. Pfeifer and P. Protzel, "Expectation-maximization for adaptive mixture models in graph optimization," in *Proc. Int. Conf. Robot. Automat.* (ICRA), May 2019, pp. 3151–3157.
- [168] W. Wen, T. Pfeifer, X. Bai, and L. Hsu, "Factor graph optimization for GNSS/INS integration: A comparison with the extended Kalman filter," *Navigation*, vol. 68, no. 2, pp. 315–331, Jun. 2021. [Online]. Available: http://onlinelibrary.wiley.com/doi/abs/10.1002/navi.421
- [169] W. Li, X. Cui, and M. Lu, "A robust graph optimization realization of tightly coupled GNSS/INS integrated navigation system for urban vehicles," *Tsinghua Sci. Technol.*, vol. 23, no. 6, pp. 724–732, Dec. 2018.
- [170] Y. Song and L.-T. Hsu, "Tightly coupled integrated navigation system via factor graph for UAV indoor localization," *Aerosp. Sci. Technol.*, vol. 108, Jan. 2021, Art. no. 106370.
- [171] R. Eberhart and J. Kennedy, "A new optimizer using particle swarm theory," in *Proc. 6th Int. Symp. Micro Mach. Hum. Sci. (MHS)*, 1995, pp. 39–43.
- [172] J. Kennedy and R. Eberhart, "Particle swarm optimization," in Proc. Int. Conf. Neural Netw. (ICNN), vol. 4, 1995, pp. 1942–1948.
- [173] X. Cai, L. Ye, and Q. Zhang, "Ensemble learning particle swarm optimization for real-time UWB indoor localization," *EURASIP J. Wireless Commun. Netw.*, vol. 2018, no. 1, pp. 1–15, May 2018, doi: 10.1186/s13638-018-1135-0.
- [174] H. Wu, J. Liu, Z. Dong, and Y. Liu, "A hybrid mobile node localization algorithm based on adaptive MCB-PSO approach in wireless sensor networks," *Wireless Commun. Mobile Comput.*, vol. 2020, Jun. 2020, Art. no. e3845407. [Online]. Available: https://www.hindawi.com/journals/wcmc/2020/3845407/
- [175] Y. Wu, S. Ding, Y. Ding, and M. Li, "UWB base station cluster localization for unmanned ground vehicle guidance," *Math. Problems Eng.*, vol. 2021, Apr. 2021, Art. no. e6639574. [Online]. Available: https://www.hindawi.com/journals/mpe/2021/6639574/
- [176] J. Khodjaev, Y. Park, and A. S. Malik, "Survey of NLOS identification and error mitigation problems in UWB-based positioning algorithms for dense environments," *Ann. Telecommun.-Annales des Télécommunications.*, vol. 65, nos. 5–6, pp. 301–311, Aug. 2009, doi: 10.1007/s12243-009-0124-z.
- [177] I. Guvenc, C.-C. Chong, and F. Watanabe, "NLOS identification and mitigation for UWB localization systems," in *Proc. IEEE Wireless Commun. Netw. Conf.*, Mar. 2007, pp. 1571–1576, doi: 10.1109/wcnc.2007.296.

- *Line of Sight Operating Conditions.* [Online]. Available: https://www.decawave.com/wp-content/uploads/2018/10/APS006\_Part-3-DW1000-Diagnostics-for-NLOS-Channels\_v1.1.pdf
   M. Elsanhoury, J. Koljonen, P. Välisuo, M. Elmusrati, and H. Kuusniemi,
  - [179] M. Eisamoury, J. Koljonen, P. Vansuo, M. Emustati, and H. Kuushenni, "Survey on recent advances in integrated GNSSs towards seamless navigation using multi-sensor fusion technology," in *Proc. 34th Int. Tech. Meeting Satell. Division Inst. Navigat. (ION GNSS)*, Oct. 2021, pp. 2754–2765, doi: 10.33012/2021.17961.

[178] Decawave. (2016). DW1000 Metrics for Estimation of Non

- [180] X. Guo, N. Ansari, F. Hu, Y. Shao, N. R. Elikplim, and L. Li, "A survey on fusion-based indoor positioning," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 1, pp. 566–594, 1st Quart., 2020, doi: 10.1109/comst.2019.2951036.
- [181] L. Guanke, Y. Jianan, Y. Wen, and Z. Yanling, "Research on seamless positioning of power wearables based on GPS/UWB combination," in *Proc. IEEE Int. Conf. Comput. Commun. Eng. Technol. (CCET)*, Aug. 2018, pp. 123–127, doi: 10.1109/ccet.2018.8542426.
- [182] Y. Wang and X. Li, "The IMU/UWB fusion positioning algorithm based on a particle filter," *ISPRS Int. J. Geo-Inf.*, vol. 6, no. 8, p. 235, Aug. 2017, doi: 10.3390/ijgi6080235.
- [183] J. F. Liu, J. X. Pu, L. F. Sun, and Z. S. He, "An approach to robust INS/UWB integrated positioning for autonomous indoor mobile robots," *Sensors*, vol. 19, no. 4, p. 950, Feb. 2019, doi: 10.3390/s19040950.
- [184] R. D. Christ and R. L. Wernli, "Navigational Sensors," in *The ROV Manual*. Amsterdam, The Netherlands: Elsevier, 2014, pp. 453–475, doi: 10.1016/b978-0-08-098288-5.00017-8.
- [185] J. Wang, Y. Gao, Z. Li, X. Meng, and C. M. Hancock, "A tightly-coupled GPS/INS/UWB cooperative positioning sensors system supported by V2I communication," *Sensors*, vol. 16, no. 7, p. 944, Jun. 2016, doi: 10.3390/s16070944.
- [186] Q. Fan, B. Sun, Y. Sun, Y. Wu, and X. Zhuang, "Data fusion for indoor mobile robot positioning based on tightly coupled INS/UWB," *J. Navigat.*, vol. 70, no. 5, pp. 1079–1097, Apr. 2017, doi: 10.1017/s0373463317000194.
- [187] J. Kim, G.-I. Jee, and J. G. Lee, "A federated Kalman filter design using a gain fusion algorithm," *IFAC Proc. Volumes*, vol. 31, no. 21, pp. 385–391, Aug. 1998, doi: 10.1016/s1474-6670%2817%2941107-4.
- [188] N. A. Carlson and M. P. Berarducci, "Federated Kalman filter simulation results," *Navigation*, vol. 41, no. 3, pp. 297–322, Sep. 1994, doi: 10.1002/j.2161-4296.1994.tb01882.x.
- [189] Y. Xu, G. Tian, and X. Chen, "Enhancing INS/UWB integrated position estimation using federated EFIR filtering," *IEEE Access*, vol. 6, pp. 64461–64469, 2018, doi: 10.1109/ACCESS.2018.2878101.
- [190] Y. Xu, C. K. Ahn, Y. S. Shmaliy, X. Chen, and L. Bu, "Indoor INS UWB-based human localization with missing data utilizing predictive UFIR filtering," *IEEE/CAA J. Autom. Sinica*, vol. 6, no. 4, pp. 952–960, Jul. 2019, doi: 10.1109/jas.2019.1911570.
- [191] H. Yang, T. Luo, W. Li, L. Li, Y. Rao, and C. Luo, "A stable SINS/UWB integrated positioning method of shearer based on the multi-model intelligent switching algorithm," *IEEE Access*, vol. 7, pp. 29128–29138, 2019, doi: 10.1109/access.2019.2898212.
- [192] S. Pala, S. Jayan, and D. G. Kurup, "An accurate UWB based localization system using modified leading edge detection algorithm," *Ad Hoc Netw.*, vol. 97, Feb. 2020, Art. no. 102017, doi: 10.1016/j.adhoc.2019.102017.
- [193] B. Hanssens, D. Plets, E. Tanghe, C. Oestges, D. P. Gaillot, M. Lienard, T. Li, H. Steendam, L. Martens, and W. Joseph, "An indoor variancebased localization technique utilizing the UWB estimation of geometrical propagation parameters," *IEEE Trans. Antennas Propag.*, vol. 66, no. 5, pp. 2522–2533, May 2018, doi: 10.1109/tap.2018.2810340.
- [194] S. Monica and F. Bergenti, "Hybrid indoor localization using WiFi and UWB technologies," *Electronics*, vol. 8, no. 3, p. 334, Mar. 2019, doi: 10.3390/electronics8030334.
- [195] A. Poulose and D. S. Han, "UWB indoor localization using deep learning LSTM networks," *Appl. Sci.*, vol. 10, no. 18, p. 6290, Sep. 2020, doi: 10.3390/app10186290.
- [196] J. Park, S. Nam, H. Choi, Y. Ko, and Y.-B. Ko, "Improving deep learning-based UWB LOS/NLOS identification with transfer learning: An empirical approach," *Electronics*, vol. 9, no. 10, p. 1714, Oct. 2020, doi: 10.3390/electronics9101714.
- [197] H. Wang, X. Wang, Y. Xue, and Y. Jiang, "UWB-based indoor localization using a hybrid WKNN-LSTM algorithm," in *Proc. IEEE 4th Inf. Technol., Netw., Electron. Automat. Control Conf. (ITNEC)*, Jun. 2020, pp. 1720–1725, doi: 10.1109/itnec48623.2020.9085050.

- [198] X. Yang, F. Zhao, and T. Chen, "NLOS identification for UWB localization based on import vector machine," *AEU-Int. J. Electron. Commun.*, vol. 87, pp. 128–133, Apr. 2018, doi: 10.1016/j.aeue.2018.02.003.
- [199] K. Yu, K. Wen, Y. Li, S. Zhang, and K. Zhang, "A novel NLOS mitigation algorithm for UWB localization in harsh indoor environments," *IEEE Trans. Veh. Technol.*, vol. 68, no. 1, pp. 686–699, Jan. 2019, doi: 10.1109/tvt.2018.2883810.
- [200] S. Djosic, I. Stojanovic, M. Jovanovic, T. Nikolic, and G. L. Djordjevic, "Fingerprinting-assisted UWB-based localization technique for complex indoor environments," *Expert Syst. Appl.*, vol. 167, Apr. 2021, Art. no. 114188, doi: 10.1016/j.eswa.2020.114188.
- [201] M. Lei, M. Jin, T. Huang, Z. Guo, Q. Wang, Z. Wu, Z. Chen, X. Chen, and J. Zhang, "Ultra-wideband fingerprinting positioning based on convolutional neural network," in *Proc. Int. Conf. Comput., Inf. Telecommun. Syst. (CITS)*, Oct. 2020, pp. 1–5, doi: 10.1109/cits49457.2020.9232628.
- [202] S. Monica and G. Ferrari, "Improving UWB-based localization in iot scenarios with statistical models of distance error," *Sensors*, vol. 18, no. 5, p. 1592, May 2018, doi: 10.3390/s18051592.
- [203] J. A. del Peral-Rosado et al., "Whitepaper on new localization methods for 5G wireless systems and the Internet-of-Things," COST Action CA15104 (IRACON), IRACON-COST, Ottignies-Louvain-la-Neuve, Belgium, Tech. Rep. COST Action CA15104, 2018. [Online]. Available: http://www.iracon.org/wp-content/uploads/2018/03/IRACON-WP2.pdf
- [204] F. Guidi, N. Decarli, D. Dardari, F. Natali, E. Savioli, and M. Bottazzi, "A low complexity scheme for passive UWB-RFID: Proof of concept," *IEEE Commun. Lett.*, vol. 20, no. 4, pp. 676–679, Apr. 2016, doi: 10.1109/lcomm.2016.2530658.
- [205] A. Costanzo, D. Dardari, J. Aleksandravicius, N. Decarli, M. Del Prete, D. Fabbri, M. Fantuzzi, A. Guerra, D. Masotti, M. Pizzotti, and A. Romani, "Energy autonomous UWB localization," *IEEE J. Radio Freq. Identif.*, vol. 1, no. 3, pp. 228–244, Sep. 2017, doi: 10.1109/jrfid.2018.2792538.
- [206] D. Shi, H. Mi, E. G. Collins, and J. Wu, "An indoor low-cost and high-accuracy localization approach for AGVs," *IEEE Access*, vol. 8, pp. 50085–50090, 2020, doi: 10.1109/access.2020.2980364.
- [207] W. Wang, D. Marelli, and M. Fu, "Multiple-vehicle localization using maximum likelihood Kalman filtering and ultra-wideband signals," *IEEE Sensors J.*, vol. 21, no. 4, pp. 4949–4956, Feb. 2021, doi: 10.1109/jsen.2020.3031377.
- [208] F. Lazzari, A. Buffi, P. Nepa, and S. Lazzari, "Numerical investigation of an UWB localization technique for unmanned aerial vehicles in outdoor scenarios," *IEEE Sensors J.*, vol. 17, no. 9, pp. 2896–2903, May 2017, doi: 10.1109/jsen.2017.2684817.
- [209] J. Zhu, J. Yao, X. Zhu, Z. Li, Y. Ma, and A. Zou, "Research on substation safety monitoring system based on UWB positioning technology," in *Proc. IEEE 3rd Int. Conf. Inf. Syst. Comput. Aided Educ. (ICISCAE)*, Sep. 2020, pp. 332–334, doi: 10.1109/iciscae51034.2020.9236864.
- [210] Y. Wang and X. Li, "The IMU/UWB fusion positioning algorithm based on a particle filter," *ISPRS Int. J. Geo-Inf.*, vol. 6, no. 8, p. 235, Aug. 2017, doi: 10.3390/ijgi6080235.
- [211] D. Tian and Q. Xiang, "Research on indoor positioning system based on UWB technology," in *Proc. IEEE 5th Inf. Technol. Mechatronics Eng. Conf. (ITOEC)*, Jun. 2020, pp. 662–665, doi: 10.1109/itoec49072.2020.9141707.
- [212] K. Minne, N. Macoir, J. Rossey, Q. Van den Brande, S. Lemey, J. Hoebeke, and E. De Poorter, "Experimental evaluation of UWB indoor positioning for indoor track cycling," *Sensors*, vol. 19, no. 9, p. 2041, May 2019, doi: 10.3390/s19092041.
- [213] F. Che, A. Ahmed, Q. Z. Ahmed, S. A. R. Zaidi, and M. Z. Shakir, "Machine learning based approach for indoor localization using ultrawide bandwidth (UWB) system for industrial Internet of Things (IIoT)," in *Proc. Int. Conf. U.K.-China Emerg. Technol. (UCET)*, Aug. 2020, pp. 1–4, doi: 10.1109/ucet51115.2020.9205352.
- [214] W. Liu, J. Li, Q. Zeng, F. Guo, R. Wu, and X. Zhang, "An improved robust Kalman filtering strategy for GNSS kinematic positioning considering small cycle slips," *Adv. Space Res.*, vol. 63, no. 9, pp. 2724–2734, May 2019, doi: 10.1016/j.asr.2017.11.041.
- [215] N. Xia and M. A. Weitnauer, "TDOA-based mobile localization using particle filter with multiple motion and channel Models," *IEEE Access*, vol. 7, pp. 21057–21066, 2019, doi: 10.1109/access.2019.2897936.
- [216] J. L. C. Villacres, Z. Zhao, T. Braun, and Z. Li, "A particle filter-based reinforcement learning approach for reliable wireless indoor positioning," *IEEE J. Sel. Areas Commun.*, vol. 37, no. 11, pp. 2457–2473, Nov. 2019, doi: 10.1109/jsac.2019.2933886.

- [217] G. M. Hoang, B. Denis, J. Härri, and D. Slock, "Bayesian fusion of GNSS, ITS-G5 and IR–UWB data for robust cooperative vehicular localization," *Comp. Rendus Phys.*, vol. 20, no. 3, pp. 218–227, Mar. 2019. [Online]. Available: https://www.sciencedirect. com/science/article/pii/S1631070519300246
- [218] A. Minetto, A. Gurrieri, and F. Dovis, "A cognitive particle filter for collaborative DGNSS positioning," *IEEE Access*, vol. 8, pp. 194765–194779, 2020.
- [219] S. Guler, M. Abdelkader, and J. S. Shamma, "Infrastructure-free multirobot localization with ultrawideband sensors," in *Proc. Amer. Control Conf. (ACC)*, Jul. 2019, pp. 13–18, doi: 10.23919/acc.2019.8814678.
- [220] S. C. Gheorghiu, K. Nagy-Betegh, R. H. Molnar, and R. C. Grammenos, "WALLSY: The UWB and SmartMesh IP enabled wireless ad-hoc lowpower localization system," in *Proc. Int. Conf. Localization GNSS (ICL-GNSS)*, Jun. 2021, pp. 1–17, doi: 10.1109/icl-gnss51451.2021.9452247.
- [221] A. Chehri, P. Fortier, and P. M. Tardif, "UWB-based sensor networks for localization in mining environments," *Ad Hoc Netw.*, vol. 7, no. 5, pp. 987–1000, Jul. 2009, doi: 10.1016/j.adhoc.2008.08.007.
- [222] T.-M. Nguyen, A. Hanif Zaini, C. Wang, K. Guo, and L. Xie, "Robust target-relative localization with ultra-wideband ranging and communication," in *Proc. IEEE Int. Conf. Robot. Autom. (ICRA)*, May 2018, pp. 2312–2319, doi: 10.1109/icra.2018.8460844.
- [223] L. Li, X. Guo, N. Ansari, and H. Li, "A hybrid fingerprint quality evaluation model for WiFi localization," *IEEE Internet Things J.*, vol. 6, no. 6, pp. 9829–9840, Dec. 2019, doi: 10.1109/jiot.2019.2932464.
- [224] A. H. Salamah, M. Tamazin, M. A. Sharkas, and M. Khedr, "An enhanced WiFi indoor localization system based on machine learning," in *Proc. Int. Conf. Indoor Positioning Indoor Navigat. (IPIN)*, Oct. 2016, pp. 1–8, doi: 10.1109/ipin.2016.7743586.
- [225] M. Nowicki and J. Wietrzykowski, "Low-effort place recognition with WiFi fingerprints using deep learning," in *Automation*. Cham, Switzerland: Springer, 2017, pp. 575–584, doi: 10.1007/978-3-319-54042-9\_57.
- [226] X. Wang and J. Shen, "Machine learning and its applications in visible light communication based indoor positioning," in *Proc. Int. Conf. High Perform. Big Data Intell. Syst. (HPBD&IS)*, May 2019, pp. 274–277, doi: 10.1109/hpbdis.2019.8735490.
- [227] C. Loffler, S. Riechel, J. Fischer, and C. Mutschler, "Evaluation criteria for inside-out indoor positioning systems based on machine learning," in *Proc. Int. Conf. Indoor Positioning Indoor Navigat. (IPIN)*, Sep. 2018, pp. 1–8, doi: 10.1109/ipin.2018.8533862.
- [228] T. Yuan, Y. Xu, Y. Wang, P. Han, and J. Chen, "A tilt receiver correction method for visible light positioning using machine learning method," *IEEE Photon. J.*, vol. 10, no. 6, pp. 1–12, Dec. 2018, doi: 10.1109/jphot.2018.2880872.
- [229] H. Tran and C. Ha, "Improved visible light-based indoor positioning system using machine learning classification and regression," *Appl. Sci.*, vol. 9, no. 6, p. 1048, Mar. 2019, doi: 10.3390/app9061048.
- [230] C.-H.-G. Li and Y.-M. Chang, "Automated visual positioning and precision placement of a workpiece using deep learning," *Int. J. Adv. Manuf. Technol.*, vol. 104, nos. 9–12, pp. 4527–4538, Aug. 2019. [Online]. Available: https://www.springerprofessional.de/en/automatedvisual-positioning-and-precision-placement-of-a-workpi/17103994
- [231] P. Sthapit, H.-S. Gang, and J.-Y. Pyun, "Bluetooth based indoor positioning using machine learning algorithms," in *Proc. IEEE Int. Conf. Consum. Electron.-Asia (ICCE-Asia)*, Jun. 2018, pp. 206–212, doi: 10.1109/icceasia.2018.8552138.
- [232] L. Zhang, N. Xiao, W. Yang, and J. Li, "Advanced heterogeneous feature fusion machine learning models and algorithms for improving indoor localization," *Sensors*, vol. 19, no. 1, p. 125, Jan. 2019, doi: 10.3390/s19010125.
- [233] R. M. Faragher and R. K. Harle, "SmartSLAM—An efficient smartphone indoor positioning system exploiting machine learning and opportunistic sensing," in *Proc. 26th Int. Tech. Meeting Satell. Division Inst. Navigat. (ION GNSS)*, Sep. 2013, pp. 1006–1019. [Online]. Available: http://www.ion.org/publications/abstract.cfm?jp=p&articleID=11308
- [234] Z. Li, K. Xu, H. Wang, Y. Zhao, X. Wang, and M. Shen, "Machinelearning-based positioning: A survey and future directions," *IEEE Netw.*, vol. 33, no. 3, pp. 96–101, May 2019, doi: 10.1109/mnet.2019.1800366.
- [235] I.-H. Li, W.-Y. Wang, C.-Y. Li, J.-Z. Kao, and C.-C. Hsu, "Cloudbased improved Monte Carlo localization algorithm with robust orientation estimation for mobile robots," *Eng. Computations*, vol. 36, no. 1, pp. 178–203, Dec. 2018.

[236] J. J. Khalife and Z. M. Kassas, "Receiver design for Doppler positioning with leo satellites," in *Proc. IEEE Int. Conf. Acoust., Speech Signal Process. (ICASSP)*, May 2019, pp. 5506–5510.



**MAHMOUD ELSANHOURY** (Graduate Student Member, IEEE) received the B.Sc. degree in telecommunications engineering from Alexandria University, Egypt, in 2013, and the M.Sc. (Tech) degree in telecommunications engineering from the University of Vaasa, Finland, in 2018, where he is currently pursuing the Ph.D. degree in computer science. From 2017 to 2018, he worked as a Research Assistant at the University of Vaasa. In spring 2017, he worked as an Autonomous

Robots Lecturer at the Swedish NOVIA University of Applied Sciences. Recently in 2021, he conducted a two-month research visit to American University in Cairo (AUC) to exchange research ideas around UWB and LEO positioning. His current research interests include ubiquitous indoor positioning systems, ultra-wideband (UWB) indoor localization, low-earth orbit (LEO) satellites for positioning, multi-sensor fusion technologies, Kalman filters, and machine learning algorithms.



**PETTERI MÄKELÄ** received the M.Sc. (Tech) degree in computer science from the Helsinki University of Technology, Finland, in 1994, and the Lic.Sc. (Tech) degree in electrical engineering from the Tampere University of Technology, Finland, in 2008. He is currently pursuing the Ph.D. degree with the University of Vaasa. He worked for more than ten years in product development of GNSS-based systems in industry before establishing a career with the Seinäjoki

University of Applied Sciences, Finland. He is also working as a Principal Lecturer with the Seinäjoki University of Applied Sciences. His research interests include positioning technologies, the industrial IoT, and software engineering.



JANNE KOLJONEN was born in Vaasa, Finland, in 1980. He received the M.S. and Ph.D. degrees in automation technology from the University of Vaasa, Finland, in 2004 and 2010, respectively.

He has been working in various research and teaching tasks with the University of Vaasa, since 2002. He is currently working as a Laboratory Engineer and a Program Manager with the University of Vaasa. He has authored more than 40 scientific articles.



**PETRI VÄLISUO** received the M.Sc. (Tech) degree in computer science from the Tampere University of Technology, Finland, in 1996, and the D.Sc. (Tech) degree in automation technology from the University of Vaasa, Finland, in 2011. He has worked for ten years in telecommunications industry before establishing a research career with the University of Vaasa. He is currently working as an Associate Professor (Tenure Track) of sustainable automation with the School of Technology and

Innovation Management, University of Vaasa. He has authored or coauthored 27 peer-reviewed and more than ten other scientific publications. His research interests include ML, the IoT, positioning methods, and other technologies relevant to industrial automation.



**AHM SHAMSUZZOHA** received the master's degree in mechanical engineering, majoring in energy and environment from the University of Strathclyde, U.K., and the Ph.D. degree in industrial management from the University of Vaasa. He worked as an Assistant Professor at Sultan Qaboos University, Oman, and an Associate Professor at the Shahjalal University of Science and Technology, Bangladesh. He is currently working as an Assistant Professor (Tenure Track) with the

Industrial Systems Analytics Program, School of Technology and Innovations and Digital Economy Research Platform, University of Vaasa. He is also involved in many European Union and nationally funded projects in Finland. His research interests include business and entrepreneurship, supply chain and logistics management, energy and environment, blockchain and precision positioning, digitalization, and green innovation.



**TIMO MANTERE** received the B.Sc. degree in electric power engineering from Vaasa Polytechnic, in 1992, and the M.Sc. degree in economics and business administration and the D.Sc. degree in computer science from the University of Vaasa, in 1996 and 2003, respectively. He is an expert in computer science and automation, mainly researching AI, ML, machine vision, signal processing, and software testing. He is currently working as a University Lecturer and an Associate

Professor of computer science and automation systems with the University of Vaasa. He previously worked as a Lecturer at the Lappeenranta University of Technology. He has published approximately 80 scientific papers, articles, and presentations.



**MOHAMMED ELMUSRATI** (Senior Member, IEEE) received the B.Sc. and M.Sc. degrees (Hons.) in electrical and electronic engineering from the University of Benghazi, Libya, in 1991 and 1995, respectively, and the Licentiate of Science degree (Hons.) in technology and the D.Sc. degree in automation and control engineering from Aalto University, Finland, in 2002 and 2004, respectively.

He is currently a Full Professor of communication, automation, and digitalization at the School of Technology and Innovations, University of Vaasa, Finland. He has more than 145 publica-

Innovations, University of Vaasa, Finland. He has more than 145 publications, including articles, books, and book chapters. His research interests include various research topics, including wireless communications, AI, ML, biotechnology, big data analysis, stochastic systems, and game theory.

Dr. Elmusrati is an Active Member in different scientific societies, such as a member of the Society of Industrial and Applied Mathematics (SIAM) and a member of the Finnish Automation Society.



**HEIDI KUUSNIEMI** (Member, IEEE) received the M.Sc. (Hons.) and Ph.D. degrees in information technology from the Tampere University of Technology, Finland, in 2002 and 2005, respectively. She has more than 18 years of experience in research and development of positioning technologies. In 2016, she was selected as one of the 50 most inspiring technology leaders in Nordic countries. Part of her Ph.D. research was conducted at the Department of Geomatics Engineer-

ing, University of Calgary, Canada. She used to be a Visiting Scholar at Stanford University's GPS Laboratory, in 2017. She is currently the Director of the Digital Economy Research Platform and a Professor of computer science at the University of Vaasa, a Research Professor at the Finnish Geospatial Research Institute, National Land Survey of Finland, and an Adjunct Professor of satellite navigation at Tampere University and Aalto University, Finland. She is also active in the 'UN Women in Space' and 'Women in Tech' networks.